

# Economics and transitions. Lessons from economic subdisciplines

### Citation for published version (APA):

van den Bergh, J., & Kemp, R. P. M. (2006). Economics and transitions. Lessons from economic subdisciplines. MERIT. UNU-MERIT Working Papers No. 2006-038

Document status and date: Published: 01/01/2006

**Document Version:** Publisher's PDF, also known as Version of record

#### Please check the document version of this publication:

• A submitted manuscript is the version of the article upon submission and before peer-review. There can be important differences between the submitted version and the official published version of record. People interested in the research are advised to contact the author for the final version of the publication, or visit the DOI to the publisher's website.

• The final author version and the galley proof are versions of the publication after peer review.

 The final published version features the final layout of the paper including the volume, issue and page numbers.

Link to publication

#### General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these riahts.

• Users may download and print one copy of any publication from the public portal for the purpose of private study or research.

You may not further distribute the material or use it for any profit-making activity or commercial gain
You may freely distribute the URL identifying the publication in the public portal.

If the publication is distributed under the terms of Article 25fa of the Dutch Copyright Act, indicated by the "Taverne" license above, please follow below link for the End User Agreement:

www.umlib.nl/taverne-license

#### Take down policy

If you believe that this document breaches copyright please contact us at: repository@maastrichtuniversity.nl

providing details and we will investigate your claim.



# **Working Paper Series**

#2006-038

# **Economics and Transitions: Lessons from Economic Sub-disciplines**

Jeroen van den Bergh

René Kemp

October 2006

United Nations University - Maastricht Economic and social Research and training centre on Innovation and Technology Keizer Karelplein 19, 6211 TC Maastricht, The Netherlands Tel: (31) (43) 350 6300, Fax: (31) (43) 350 6399, e-mail: info@merit.unu.edu, URL: http://www.merit.unu.edu

# **Economics and Transitions:** Lessons from Economic Sub-disciplines<sup>1</sup>

Jeroen van den Bergh Faculty of Economics and Business Administration & Institute for Environmental Studies, Free University De Boelelaan 1105 1081 HV, Amsterdam jbergh@feweb.vu.nl

and

René Kemp UNU-MERIT Maastricht University Keizer Karelplein 19 6211 TC Maastricht r.kemp@merit.unimaas.nl

## Abstract

Currently, there is much interest in stimulating or 'speeding up' socio-technical transitions to sustainable systems, most notably in the sectors of energy, transport and agriculture. This essay attempts to assess whether and how 'transition' type problems and issues are being addressed in the various sub-disciplines and methodological approaches of economics. This allows us to identify concepts, ideas, theories and empirical methods in economics that are suitable for inclusion and elaboration in 'transition research'. Surprisingly, we find that many sub-disciplines of economics have in one way or another addressed problems similar to transitions. Our main conclusion therefore is that economics offers a rich palette of ideas that may be useful for transition research. Studies on development stages, long waves, technological path-dependency, conflict resolution, public investments, emergence of institutions and, transitions from communist to market-democracy systems seem especially relevant to the study of transition. Although mainstream economics conflicts in certain ways with the approach called for by many involved in transitions.

# UNU-MERIT Working Papers ISSN 1871-9872

UNU-MERIT Working Papers intend to disseminate preliminary results of the research carried out at the institute to attract comments

<sup>&</sup>lt;sup>1</sup> This paper was commissioned by the *Kennisnetwerk Systeem Innovatie en Transities* (KSI) with the aim to gain insight into economic perspectives on societal transitions and their management. We thank John Grin, Jan Rotmans and Johan Schot for comments and suggestions, and Maria Healy for editorial support.

# **1. Introduction**

Recent discussions of sustainable development often present the concept of 'transition' as a generalized notion of system innovations (e.g., Board on Sustainable Development, 2000; Raskin et al., 2002; Elzen et al., 2004; Geels, 2002, 2005; Rotmans, 2003; Smith et al., 2005). A transition comprises a society-wide change that goes beyond single sectors and involves fundamental and interrelated changes in technology, organisation, institutions and culture. The notion of 'transition' can be viewed as a response to the problem associated with the fact that many studies of sustainable development are restricted to formulating or portraying a hypothetical and utopian sustainable system (i.e., a 'blueprint') while it remains unclear how such a system can be realized. The shifting of the focal point from a sustainable 'end state' to the transition process that transforms the current, unsustainable system into a sustainable state, adds realism and policy relevance to the analysis. Transition studies are thus aimed at assessing the processes that initiate, foster and direct transitions as well as the barriers against transitions.

In the Netherlands, the national government has made a commitment to stimulate transitions in energy, agriculture, built environment and transport. In addition, a 20 million  $\in$  research programme (KSI) has been established to increase knowledge about transitions. The programme aims to generate both fundamental knowledge about transition processes, including typologies of transitions, causal mechanisms, and ways in which transitions and related co-dynamic processes can be studied. In addition, it aims to produce practical knowledge that can be used to the benefit of ongoing transition experiments in the Netherlands, which are fostered by the ministries of economics and the environment. Both aims are realized through three areas of research:<sup>2</sup>

- Historical research of past transitions, which allows drawing theoretical as well as policy lessons. In particular, this tries to learn about the teleological nature of past transitions, the role of co-dynamics of sub-systems, visions, expectations, and collective action in bringing about transitions.
- 2) Systematic analysis and monitoring of current transitions and analysis of hypothetical, future transitions, with a special focus on mobility, health care and, agriculture.
- 3) Research on the governance (steering and management) of transitions, looking at the codynamics of sub-systems.

Transitions can be conceptualised as non-linear processes of change, involving four stages, namely predevelopment, take-off, acceleration or breakthrough, and stabilisation (Rotmans, et al., 2001). Steering of transitions is difficult if not impossible, as they involve complex interactions between subsystems (co-dynamics) of the social-economic-technological-institutional system. Even when a government commits itself to a certain transition, it cannot control the transition in a top-down manner, if only because it itself is a central part of the change.

Kemp et al. (2006) have developed a model of transition management based on such things as 'dynamic agendas' and 'adaptive programmes'. Transitions are regarded here as involving changes at three levels, namely of niche, regime and (socio-technical) landscape. A transition requires synchronicity and interaction of developments in different domains, which often involves mutual reinforcement (Rotmans, et al., 2001). Within the multilevel scheme, producers are part of a production regime and a technological regime based on a certain body of knowledge with links to science and tacit elements. Users are part of a user regime, characterised by income, (changing) preferences, habits, capabilities and modes of social interaction. They are constituted (configured) by certain capabilities, values, beliefs and roles. In addition to producer, user and, technology regimes, there are regimes of policy, science, socio-cultural activities and, market functioning. Developments in one regime influence developments in another regime

<sup>&</sup>lt;sup>2</sup> For an overview of KSI projects see <u>http://www.ksinetwork.nl/?content=projects</u>.

(Geels and Kemp, 2005), which leads to co-dynamics or even coevolution (see Section 3.2). Niches are domains where radical innovations emerge. In other words, they act as 'incubation rooms' for radical novelties, shielding them from mainstream market selection, acting as stepping stone for further change (Schot, 1998). The macro-level is formed by the socio-technical landscape, which refers to aspects of the exogenous environment. This includes macroeconomic and international conditions, politics, cultural and normative values, environmental problems and scarce resources. It further includes the material and spatial configurations and arrangements of cities, factories, and physical (road and energy) infrastructure (Rip and Kemp, 1998; Geels, 2002, 2005).



*Figure 1. Interaction between different scale-levels Source:* Geels (2002).

There are different types of transitions. History depicts both relatively large and small transitions. Examples of large transitions include the invention and spread of fire control, the rise of agriculture, the Industrial Revolution and, the emergence of mass-production/consumption. Smaller transitions are the Green Revolution in agriculture, the electrification of society, various transitions in transportation (e.g., horse/wagon to car) and, certain energy transitions (e.g., from coal to gas). The notion of (a small) transition may also be regarded to cover transformations in human communication (e.g., mobile phone, email, internet), and work organization and manufacturing (i.e., from handcraft to mass manufacturing and flexible production). Finally, transitions have spatial dimensions or repercussions: cities and regions go through transitions as part of wider, global transitions; associated with this, trade patterns between regions and countries change. Kemp and Rotmans (2005) distinguish two types of transitions: goal-oriented transitions and evolutionary (non-teleological) transitions. Geels and Schot (2005) make a distinction between five transition pathways, namely transformation, opening up of new functional domain, technological substitution, de-alignment and, re-alignment. In addition, one can identify a number of challenges in analyzing transitions. See Table 2 in Section 4 for further illustration.

To date, economists have not contributed much to the elaboration of the notion of 'sustainability transitions'. Nevertheless, they traditionally have devoted much attention to the study of long term growth and development as well as to the (optimal) planning of these. In addition, they have developed policy theories based on incentive mechanisms, which have also had wide application to environmental and resource regulatory problems. Economists have stressed that in many (but not all) circumstances decentralization through price incentives gives the best outcomes from a social welfare perspective. The various mainstream economic theories have not directly addressed transition challenges like path-dependence, lock-in, diversity of options, bounded rationality and, uncertainty. In this sense, there is a gap and even a conflict

between mainstream economics and the transition paradigm. Although we pay attention to this possible conflict, the current paper adopts a constructive approach by examining which useful ideas and suggestions for transition research can be derived from the rich literature on economics, both within mainstream (neoclassical) and heterodox economic schools. In particular, we will try to collect the most exciting ideas from a range of possibly relevant sub-disciplines of economics. In addition, we will examine the relevance of methodological approaches common in economics to address transition issues.

Even though the specific notion of transition is not used much in economics, several economic concepts and terms bear close relationship to it. Examples are development, growth, structural change, (system) innovation and, transformation. So far, the term 'transition' has appeared in only three areas of economic research. In theoretical, dynamic models of growth, a transition is sometimes used to denote the change from one state to another stationary (equilibrium) state. This resembles some of the interpretations commonly attached to 'transition' in the context of sustainable development, notably the association with a major technological innovation. A second area in which the notion 'transition' has been used is the study of processes characterizing the transformation from a planned to a market economy (notably, the former communist USSR and Eastern-European countries). A third, somewhat related field of research in which the term 'transition' is employed, is that of development economics, which analyses the conditions under which poor, rural countries can change into modern market-based economies, characterized by a more educated labour force, a more balanced and elaborate sector structure, with well-functioning markets, and less dependence on resource exports and foreign aid by rich countries.

This illustrates that the idea of transition is not entirely new. Moreover, transitions are not hypothetical constructs: history is full of them. This suggests a potentially important role for (economic) historical analyses. The main difference between many historical transitions and the ones envisioned in the context of a sustainable development is that whereas the first type are with few exceptions autonomous and unintended, the latter arguably require purposeful public guidance and interference. We will suggest here that the discipline of economics can provide useful insights about the latter. This is not surprising perhaps in view of its long-standing tradition of dealing with concrete policy questions – witnessed by, among others, economic sub-disciplines such as environmental and public economics.

The structure of the paper is as follows. Section 2 assesses the main methodological approaches and sub-disciplines of economics that contain potentially relevant ideas on transitions. Next, Section 3 presents a selection of the most promising approaches and sub-disciplines with a brief discussion of their core notions and insights. Section 4 summarizes the insights by linking transition challenges to economic theories and concepts. Section 5 discusses the potential conflict between mainstream economics and the 'transition governance paradigm'. Section 6 provides a summary and conclusions.

#### 2. An overview of relevant economic approaches and sub-disciplines

This section attempts to assess various ideas (i.e., concepts, theories, approaches and insights) from the broad spectrum of economics that have the potential to benefit the recent, emerging thinking on transitions. Such ideas may contribute to an improved understanding of transition phases and processes, elements of transition management, and barriers to transitions. For this purpose, we identify two levels of organisation within the discipline of economics.

The first level of organisation focuses on the *methodological starting point*. This includes:

- neoclassical (micro)economics (including partial and general equilibrium theory and game theory),
- macroeconomics (including various 'schools' of thought, such as new and post-Keynesian economics, new classical economics, supply side economics, disequilibrium theories and monetarism),

- classical economics (internally heterogeneous linked especially to early economists such as Adam Smith, Thomas Malthus, David Ricardo, Karl Marx and John Stuart Mill), and,
- various (other) heterodox schools (e.g., behavioural economics, evolutionary economics, Austrian economics, Sraffian economics and institutional economics).

The second level of organisation focuses on the *sub-disciplines*. The following list outlines the sub-disciplines considered most relevant for the purpose of transition research (core issues and concepts are indicated in brackets):

- business cycle theory (long waves),
- development economics (micro, macro, industrialization, institutional arrangements, international linkages, development planning, trade policy),
- economic growth theory (endogenous growth, multisectoral growth, comparative studies of countries),
- economic history (general, technological, industries, geography and regions, institutions),
- economic studies of transition economies (former socialist/communist countries),
- economics of disasters (disaster as unlocking a system),
- economics of information (decision-making under uncertainty and imperfect or asymmetric information, learning theories, search, network formation),
- economics of technology (innovation systems, positive externalities, diffusion, intellectual property rights, subsidies),
- environmental and energy economics (regulation, resource curse or Dutch disease, resource pricing, and extraction),
- industrial economics (market structure, innovation strategies, rent seeking, cooperation, complementary products, networks, vertical integration, and demography of industries),
- international economics (trade, location, cooperation, innovation and international diffusion),
- marketing (life cycle of products, and market and innovation strategies),
- population economics (interaction between individual choices, economic development and demography),
- public economics (public goods, public investment, public R&D, infrastructure, club theory, public and social choice, conflict resolution), and,
- spatial economics (regional diversity, spatial isolation, spatial diffusion, agglomeration theory, new economic geography, economic effects of infrastructure).

In the next section we provide a brief overview of core aspects for a select number of methodological approaches and sub-disciplines listed above, judging each against the context of the potential for transitions. Section 3.1 focuses on methodology and, Section 3.2 on sub-disciplines.

# **3.** A selection of economic insights with potential relevance for transition research **3.1** Methodological approaches

# Classical economics

Classical economists' concern with historical change suggests relevance for transition research. They study the mechanisms of transformations occurring in economic and social life. Their economic analysis is mixed with sociological analysis of social relations (struggle). The multidisciplinary nature of this approach to analysis explains why early economists like Malthus and Marx are considered 'early' sociologists by those working in sociology.

David Ricardo and John Stuart Mill regarded the economy as moving towards a steady state. From Ricardo's point of view, this was due to diminishing returns to increasingly marginal land used for agriculture, and from Mill's point of view, it was due to diminishing returns to utility. Daly (1977/1991) emphasized the notion of a steady state economy, later referred to as an

environmental sustainable economy, with a minimal material and energetic throughput and a stable population. Malthus believed population control was the major problem, as food scarcity would become urgent, and as such, he proposed restraint as the most effective and realistic approach. Classical economists generally believed that land was (still) an important and scarce production factor (Hubacek and van den Bergh, 2005). This changed dramatically due to industrialization, which marginalized agriculture. The new point of view was reflected in the neoclassical economics paradigm by a shift in emphasis to the production factors, labour and capital, resulting (initially) in a disregard for the fundamental role of natural resources.

#### Grand development theories: Marx, Schumpeter and Rostow

Classical and other economists have proposed grand development theories. Karl Marx, Joseph Schumpeter and Walt Rostow (1960) are the most important ones. They combined an interest in theory with an interest in history, not just economic history but also social and political history. They did not focus on economic change at the margin but on how different elements of historical change are mutually connected and on the discontinuities rather than trend-like changes. Marx and Schumpeter both believed that capitalist systems would go through a transition ending in a socialist system. Marx developed a political transition theory with a period of proletarian dictatorship in the transition from capitalism to socialism. Class struggle (a sort of group theory) was a crucial mechanism in his theory.

Schumpeter (1934, 1939, 1942) thought that the neoclassical notion of equilibrium was problematic, as he regarded capitalism as a process that can never be stationary. Schumpeter's view of capitalism was built around the notion of "creative destruction", denoting a "process of industrial mutation (...) that incessantly revolutionizes the economic structure *from within*, incessantly destroying the old one, incessantly creating a new one" (original italics, Schumpeter, 1976, p. 83).

Like Marx, Rostow (1960) developed a multi-stage theory of development, not towards a communist society but towards a consumer society. His five-stage framework described the stages that poor, developing countries must pass through in order to become modern market economies. These stages comprise the traditional society, the preconditions for take-off, the actual take-off, the drive to maturity, and high mass consumption. This theory resembles lifecycle theories common in marketing, aimed at describing the development of products and markets. Rostow set out a number of conditions that were likely to occur in investment, consumption and social trends during each stage. The stages, the length of transitions periods and the conditions may vary from country to country and region to region. Rostow attached great importance to political change, most notably the building of an effective national state and commitment to modernization, as an important precondition for take-off. Maturity (mass consumption) is reached after some 60 years after the beginning of take-off. The impetus for growth comes from new technology, diffusing across various sectors, ultimately leading towards the production of consumer goods and services as leading sectors. Rostow's systematic, detailed theory certainly deserves serious attention from transition researchers. A similar stage model of industry development around technology evolution focusing on skills, investment and location is offered by Perez and Soete (1988). Today however many economists are criticial about the view of economic development moving towards stationary states, not just neoclassical economists but also evolutionary economists such as Harvey and Metcalfe (2005) who regard capitalistic systems as restless and unsettling.

#### Neoclassical economics

The dominant methodology within the current discipline of economics is neoclassical economic theory. Central to it is the assumption that agents – households, firms and public agencies – act rationally or individually optimally (i.e. they consistently maximize a given utility or profit function). Neoclassical economics focuses its attention on market relationships, as such its policy

theory derives from the idea that market failures, causing a deviation between market equilibrium and a social welfare optimum, should be appropriately corrected. Suggestions resulting from this approach that are relevant to the understanding of transitions can be summarized as follows (see also den Butter and Hofkes, 2004):

- Prices (neglected in many perspectives on transitions) coordinate economic choices, including demand, supply, investment in capital and outlays on R&D.
- Negative externalities due to environmental pollution and positive externalities of R&D and innovation need to be internalized.
- Changes take the form of smooth trends (growth theory, rational expectations) and smooth substitution process (equilibrium theory) rather than sudden breaks. Economic change is governed by prices signalling relative scarcity, including those of natural resources.
- Old, obsolete techniques are gradually replaced by new, more profitable techniques (vintage capital models).

Traditional (general) equilibrium theory is a special case of game theory, where the latter generalizes types of human interactions, not only market, possibly repeated, but strategic (expectations about others' behaviour). Transitions can be conceptualised as non-zero sum, dynamic (sequential) and, coordination games. A number of insights from game theory relevant to transitions follow. A central insight of game theory is the prisoner dilemma: individual rationality combined with non-cooperation can lead to an outcome that is sub-optimal from an individual and a social perspective. Repeated interaction, negotiation (a convention, standard or institution) or coordination (information sharing, or exchange and complementary specialization), can assist in solving this type of problem. Coordination games can be characterized by multiple equilibria. Any observed or resulting equilibrium is then a matter of arbitrariness or randomness, but may be purposefully stimulated by agreeing on norms or setting uniform standards. Evolutionary game theory has been developed partly in response to this 'equilibrium selection' problem. Transitions involve many strategic interactions between different agents which might be caste in game theoretical terms.

Many of the specific insights arise from the application of general neoclassical economic theory in sub-disciplines, as examined in the next section. In addition, in Section 5, we will consider the potential conflict between mainstream neoclassical economics and transition thinking, as noted by some transition researchers.

#### Behavioural economics

Whereas neoclassical economics assumes that individual agents are rational, a new school, behavioural economics, works along different lines. Behavioural economics, overlaps with and is motivated by experimental economics and economic psychology. The three approaches have pointed convincingly at the shortcomings of the neoclassical economics approach on the basis of philosophical and theoretical arguments as well as empirical and experimental evidence (e.g., Caldwell, 1984; Conlisk, 1996). In response to this, a range of alternative models have been proposed to address particular aspects of bounded rationality (Camerer et al., 2003). Transition research calls for studies addressing barriers to change, including those relating to individuals and organizations making decisions characterized by bounded rationality

Herbert Simon was the first to systematically argue that the combination of imperfect and costly information and limited capacity of the human brain implies a procedural rather than a substantial type of rationality, resulting in 'satisficing' behaviour: individuals try to attain acceptable rather than optimal levels of welfare, profit or other indicators. From a 'hierarchy of needs' (Maslow) or lexicographic preference perspective it is argued that needs have a hierarchical order. For instance, higher needs, such as the desire for music, would not appear before the lower needs, like satisfying hunger, are satisfied. Empirical happiness or subjective

well-being research has further established that humans adapt to changing circumstances (preference drift) and value relative (income) positions and status goods (reference drift, rivalry in consumption) (van Praag and Ferrer-i-Carbonell, 2004). Imitation (bandwagon effect) is related to this, most notably with regard to conspicuous consumption or status goods. In addition, our social behaviour is complex. Penn (2003) notes that environmental regulation should, for instance, take social interactions (e.g., reputation effects) into account. This is consistent with the findings of economic psychology and experimental economics, and of group selection theory (Henrich, 2004). Group phenomena can be linked to a variety of social – non-selfish or other-regarding – preferences: reciprocal fairness, inequity aversion, pure altruism, altruistic punishment, and spite or envy (Fehr and Fischbacher, 2002).

Various theories deal with behaviour under uncertainty, prospect theory being the most influential one. Motivated by experiments, it stresses the asymmetry with which individuals perceive gains and losses. 'Habits and routines' are often regarded as a straightforward approach to deal with complexity and uncertainty. Finally, according to some theories, under highly uncertain conditions behaviour takes the form of imitation, an example of which is panic selling. Such imitation leads to a reduction in the diversity of individual behavioural strategies.

These various views on individual behaviour are relevant to transition research, as they provide clues to such different issues as imitation on the demand side in relation to lock-in, emergence of niche markets, investment and R&D decisions, strategic interaction of individuals, and relevant policy goals. For example, happiness research has shown that growth of individual welfare is not always equivalent to growth in consumption or income, if only because the rivalry game for status goods is at best a zero-sum game (Layard, 2005).

#### Evolutionary economics

Evolutionary economics is perhaps the most internally consistent and formalized alternative theory to neoclassical economics. Its starting point is formed by one or more populations of elements (agents, strategies, organizations, institutions or technologies) which are characterized by internal diversity (heterogeneity). The latter can change in two directions: it can increase through innovations and decrease through selection (adoption, imitation, diffusion) (Nelson and Winter, 1982). The model of evolution points to the cumulative nature of these changes and the associated (mostly) gradual adaptation of individual elements to their environment (possibly changing) which is comprised of economic, social and ecological conditions. Evolution thus represents a population dynamics approach to transitions.

Potts (2000) has formulated an interesting proposal for the future direction of evolutionary economics. He presents a type of axiomatic foundation for evolutionary economics. In his view, economic systems are complex 'hyperstructures' (i.e., nested sets of connections among components). Connections can represent the physical connections between components in products or machines, as well as the material and information flows between individuals or departments within an organization. In this case, economic change and growth of knowledge are in essence the process of changes between connections. One thing is evident: new products, new firms and new sectors rise and old ones disappear, while firm 'growth' and economic 'growth' are essentially processes creating new and losing old connections, as well as grouping those connections or hyperstructure dynamics. In line with the idea of changing connections, Potts calls for a new microeconomics based on the technique of discrete, combinatorial mathematics, similar to graph theory, to study the change of microeconomic connections. In addition, Potts' approach can be viewed as a fundamental discussion of the need for multi-agent or population models, also known as 'artificial life' models. He also seems to suggest that all connections have a spatial dimension as well, implying the relevance of the 'geometry of space'.

The emergence of new levels of reality through the grouping of connections offers a refreshing perspective on transitions, similar to the way in which they are perceived in biology (e.g., from molecules to cell to multicellular organism to animal groups). Not surprisingly,

transitions are also recognized and given much attention in evolutionary biology (Maynard Smith and Szathmáry, 1995). The emergence sequence suggests increasing complexity fed by a larger degree or a higher level of specialisation, labour division and cooperation. This process might be the key to fostering transitions. The essence of evolution is that its' transitions result from selforganisation based on evolutionary principles and that regulation can only guide the evolutionary mechanisms rather than control a transition.

Next, the concept of coevolution provides a promising direction for research on transitions. In a strict sense, coevolution occurs when interactive populations evolve subject to interactions taking the form of mutual selection. In other words, two evolutionary processes are interlinked or interdependent (van den Bergh and Stagl, 2003). The resulting dynamics can be complex (irregular or even seemingly chaotic) and unpredictable (e.g., Noailly, 2003). Coevolution should be clearly differentiated from co-dynamics, which is, the interaction (e.g. negative and positive feedback) between subsystems in a larger system (van den Bergh and Stagl, 2003; Winder et al., 2005).

Attempts to elaborate the notion of coevolution in an environmental policy setting are Rammel and van den Bergh (2003), Rammel et al. (2004) and Bleischwitz (2003). A coevolutionary view is important for thinking about transitions and governance for two important reasons (Kemp et al., 2005). First, it accepts that we have cause-effect-cause loops across different scales and systems, with effects becoming causes of other developments. For example, people's needs are partly endogenous to other developments (e.g., in transport and information/communication technology). This creates irreversibilities. Second, a coevolutionary perspective sees developments in different subsystems as partially independent (relative autonomy). This means that transitions cannot be managed from the top. Transitions typically evolve, with a limited role for comprehensive control. Finally, group selection may be relevant to the understanding of transitions, as it can explain group specific norms and group formation and conflict. According to group selection theory, selection is a multilevel process, occurring both within and among groups. The outcome of this depends on the relative force of each selection mechanism. If group selection effects are sufficiently strong, then groups are adaptive. This theory is used to explain altruism, cooperation, public goods and norms in groups of sizes beyond which kin selection and (direct or indirect) reciprocal selection can work. Although traditionally strongly debated in biology, the possibility of group selection is now supported by a range of theoretical models (Bergstrom, 2002). The empirical relevance of the theory differs amongst species, while its relevance for humans has recently been convincingly argued for (Wilson, 2002). A distinction can be made between genetic and cultural group selection, representing different transmission mechanisms that are operative, and which have distinct features and speeds (see a special issue of the Journal of Economic Behavior & Organization vol. 53(1), 2004). For example, in cultural group selection, higher institutions such as religions and prolonged education systems may influence the basic norms of individuals (notably children). In addition, non-random assorting (joining or forming groups based on similar individual characteristics or convictions) is more common in human social-economic systems. Group size, formation and stability are critical issues when examining group selection. They may provide useful policy linkages when applying the theory, in the context of transitions, to cooperation, group competition and stakeholder groups.

#### Institutional economics

Institutional economics is concerned with the identification of the various institutional mechanisms that coordinate economic activity, with 'getting into grips' with the circumstances under which these various mechanisms emerge, and with the logic inherent in the different coordination mechanisms (Hollingsworth and Boyer, 1997). Institutional economics developed as a separate branch of economic theory. Within it, a common distinction is made between old

(Veblen, Mitchell, Commons) and new streams (Myrdal, North, Olson, Williamson) (see Hodgson, 1988).

Mainstream economics has generally regarded institutions as constraints that can be altered by purposeful policy and regulation. In other words, institutional change is commonly framed as a control problem rather than an endogenous phenomenon. An exception to this general rule is found in the Coase theorem which states that in the presence of negative externalities, spontaneous negotiations among rational agents can lead to socially optimal outcomes. Applied institutional economic analysis has focused on the liberalization and the removal of imperfect competition in markets (anti-cartel legislation). In addition, the creation of well-functioning new markets has received some attention, most notably in the context of environmental policy (tradable permits) and semi-public goods (telecom auctions).

The two approaches in institutional economics differ in terms of their theories of change: for 'new' institutional economists, the pursuit of economic efficiency (through competition and political struggle) causes institutions to change. North (1981, 1991) has interpreted long periods of change in these terms. The other approach (i.e., 'old' institutional economists) is less economic in its identification of mechanisms. It emphasizes the historical nature of change, the role of contingency and the non-universalistic element in micro-change. Besides markets the following modes of governance (forms of interaction and mechanisms for compliance) are distinguished: communities, networks, associations, private hierarchies and the state. The modes differ in terms of organizational structure, rules of exchange and (individual and collective) means of compliance. Networks are based more on personal relationships and trust built outside of the economic arena. Private hierarchies and the state are structures that are more hierarchical structures, based on coercion and prescriptive rules (Hollingsworth and Boyer, 1997, p. 15-16). Institutions shape but do not determine behaviour. Historical change is both the result and determinant of institutional change (duality of structure).

Institutional economics is concerned with transaction costs and trust (social capital) which are seen as important variables for explaining the link between micromotives and macro-behaviour – being an important transition topic. Institutional analysis is applied to organizations, economic sectors, values chains, and nations - all of which are relevant to transition research.

#### (Neo-)Austrian economics

The school of (Neo-) Austrian economics emphasizes the roundabout, multi-stage nature of production, temporal aspects of production, production processes as compositions of discrete techniques, and market processes instead of equilibrium. Both (Neo-)Austrian economics and evolutionary economics support a type of Schumpeterian competition of a different production process: an old technique is threatened and slowly replaced by a newer, more advanced, and usually more roundabout technique (i.e. involving more indirect or intermediate production connections). Such a multi-stage and multi-activity approach clearly fits within the study of transitions.

Faber and Proops (1990) propose a neo-Austrian approach with evolutionary elements, to emphasize the role of time. They allow for irreversibility of changes in the sector structure of the economy, for uncertainty and novelty, and for a teleological sequence of production activities ("roundaboutness"). The long-term relation between environment, technology and development is characterized by the following three elements:

- The use of non-renewable natural resources is irreversible in time, therefore a technology based on its use must ultimately cease to be viable.
- Inventions and subsequent innovations lead to more efficient use of currently used resources and substitution by resources not used previously.
- Innovation requires a certain stock of capital goods with certain characteristics is built up.

Faber and Proops construct a multi-sector model with the production side formulated in terms of activity analysis, which allows for the study of the effect of invention and innovation in moving from a situation with simple production activities to more complex or roundabout production activities. Roundabout activities use multiple technologies. For example, food production has become more roundabout, moving from agriculture with labour, through agriculture with labour and capital, to a large food processing industry with many intermediate deliveries. This approach is extended with the technology effects of resource scarcity as indicated above. It can then simulate economic and environmental history from a pre-industrial agricultural society to an industrial society using fossil fuels and capital. It allows for a combination of continuous changes in technological efficiency and discrete jumps in the number of sectors and interdependencies among sectors. With its original multi-stage perspective – an alternative to neoclassical economics – on structural economic change, neo-Austrian economics could provide a useful contribution to transition research.

Some other methodological approaches, notably macroeconomic growth theory, will be addressed in the next section.

#### 3.2 Sub-disciplines

## Business cycles and long waves

The phenomenon of long waves is relevant in the context of transition research. Long waves can be defined as cycles of prices, wages, outputs of specific basic commodities (e.g. energy resources, metals), foreign trade, interest rates, and various other economic variables. The notion of waves or cycles suggests an upward and downward swing, rise and decline (boom and depression).

Various types of cycles or waves have been identified (cf. Freeman, 1996) including the Kitchin cycle (40 months) related to keeping inventories; the Juglar or business cycle (7-11 years) related to adjustment of investment in fixed assets responding with delays to price changes; the Kuznets cycle (15-30 years) related to waves of migration and weather (exogenous ''luni-solar tides' affecting rainfall and, in turn, crop production); and, the Kondratieff cycle (40-60 years). The Kondratiev growth cycle is perhaps the most interesting from a transition point of view because it is based on clusters of innovations.

Many different opinions have been expressed regarding the nature of long Kondratiev waves as well as their causes (Freeman, 1996). They are often regarded as being caused by major shifts in technology, also known as the emergence of a new techno-economic paradigm, which are followed by the diffusion of pervasive technologies among many sectors. Good examples are electricity and ICT. Freeman points out that the issue of long waves is contentious even among those who recognize fluctuations in economic variables over time. A deep methodological problem arises from the combination of the complexity of long-term history and the difficulty of empirically assessing the precise causality behind the long-wave phenomena. Reconstruction of historical data and statistical 'de-trending' (cliometrics) magnify these problems. Nevertheless, there is much to learn for transition researchers.

Freeman and Perez (1998) identify five techno-economic transitions or Kondratieff waves: (1) early mechanisation, (2) steam power and railway, (3) electrical and heavy engineering, (4) Fordist mass production, and (5) information and communication technology. For each of the paradigms they identified the main carrier branches and induced growth sectors, the key factor industries offering abundant supply at decreasing prices, the sectors growing quickly from a small basis, the limitations provided by previous techno-economic paradigms, the ways in which the new paradigm provides solutions to certain problems, and finally the organization of firms and types of cooperation and competition. Similar to Rostow, Freeman and Perez speak of a "period of transition" to denote the deep structural change in the economy requiring an equally profound transformation of the institutional and social framework (Freeman and Perez, 1988, p. 59).

#### Development economics

Development economics is concerned with barriers to and conditions for transitions from informal economies dominated by subsistence agriculture and local autarky to formal market economies involving market exchange, international trade and industrial activity. As a result, this field has many insights to offer on transitions, which would require a separate paper in itself to summarize. An important model is the one already discussed in the previous section by Rostow. Two other important ideas from development economics are backward linkages (Hirschman, 1958) and cumulative causation (Myrdal, 1957), topics which are being revisited by innovation economists in models of endogenous growth. Countries may get stuck in a low-growth equilibrium due to specialization in low-productivity techniques and sectors that generate few knowledge spillovers. Processes of catching up and falling behind have been studied by Fagerberg (1994) and Fagerberg and Verspagen (2002) in models of technological gaps where a gap that is too large inhibits the process of catching up. Dynamic processes associated with the way in which an equilibrium gets established have also been analyzed in expectation-based models where the relative importance of the past and the expected future is shown to depend on parameters like the discount rate and the speed of adjustment (Pardham, 1995).

The "history versus expectation" controversy is not resolved (Krugman, 1991). It is generally agreed that any type of coordination for higher growth is not an easy task. The role of government is being reassessed within the field of development economics: government policy is increasingly viewed as holding back development rather than promoting it, due to government failures. The role of the government escaping development traps continues to be debated. Whatever the government's role is, it is not simple, in part because development involves structural change and newly created capabilities that are difficult to manage. The use of foreign technology is not a real solution because of "tacitness and circumstantial sensitivity" of technology (Pardham, 1995). One lesson taken from development economics is that development defies precise control. Another is that it is difficult to manage adjustment processes, the interplay between macro-economic variables, and microeconomic processes. From development economics, transition research may borrow from the method of comparative analysis (of institutional arrangements such as good governance) and models of low-equilibrium traps (see Hayami, 1998).

There are several other relevant ideas within development economics. The Kuznets curve, an inverted U-shaped relationship between income per capita (horizontal axis) and income inequality (measured by, e.g., a Gini coefficient) formalizes that agricultural economies have a low level of income inequality, during early industrialization income inequality increases, while beyond a turning point inequality decreases (Kuznets, 1955). In an analogous manner, the notion of an environmental Kuznets curve (environmental pressure versus income per capita) has been examined, but has been shown to provide a less general result (de Bruyn and Heintz, 1999). Kuznets was also one of the first to suggest that the current developing countries were very different from what the western countries were before the time of industrialization. In other words, he stresses that the 'linear model' according to which all countries go through the same stages, is incorrect. Kuznets (1971) argued that various conditions are needed for development. Advancing technology is necessary but insufficient. Institutional and attitudinal adjustment which might be called 'social innovation' – are also required. A number of changes characterize the development process: increases in average per capita income, total factor productivity and international trade and relations and, structural transformation of the economy, as well as social and ideological transformations.

There are many other concepts, proposals and models in development economics that are potentially relevant to transition research: structural change models (e.g., Lewis' 2-sector model of food production and manufacturing), international-dependence models (neocolonialism), falseparadigm model (not accounting for unique cultural, tribal, caste and institutional circumstances), dualistic-development thesis (combination of modern and traditional approaches; of wealthy, educated elites and masses of illiterate, poor people; and of monetized markets and informal barter trade), coordination failures (market failures, lack of complementary and intermediate sectors, lack of specialized labour – e.g., O-ring theory), differing initial conditions (physical resources, human resources, climate, population size), neoclassical economics perspectives on market liberalization and internationalization, analysis of a range of other factors (demography, quality-quantity trade-off with regard to offspring, migration, foreign aid). See, for instance, Todaro and Smith (2003) for an introduction to these various ideas. It is therefore likely that most of the ideas currently generated by transition researchers are reinventions of older ideas by previous researchers of economic development.

#### *Economic growth theory*

Within neoclassical economics, the theory of economic growth offers the most clear and strong link with transitions. This theory includes a variety of approaches. One can distinguish between classical (Smith to Marx), Keynesian (Harrod-Domar, Meade), neoclassical (exogenous; Solow-Swan), endogenous growth theory (Romer, Lucas), and others (e.g., von Neumann). In addition, it is possible to identify positive and normative theories (Tinbergen, Cass-Koopmans, Ramsey, golden rules, turnpike), and multisectoral approaches (Uzawa). Policy relevant issues relate to convergence of growth rates and levels among countries, avoidance or undoing of poverty traps.

Transitions are characterized by non-linear developments. Transitions have periods when development is slow and at other times, fast. In theoretical dynamic models of growth theory, these features of a transition are recognized without having to specify micro-processes. In fact, 'transition' is used sometimes to denote the shift to a higher equilibrium growth path. In normative growth theory, transitions are associated with the 'turnpike' theorem. This theorem states that it can be optimal from an intertemporal welfare perspective to follow a (close to) maximally rapid growth path to allow an economy to move to a more satisfactory state quickly, despite consumption being lower during the transition than at the beginning or end of the period (Dorfman et al., 1958).

Stability is another aspect of growth theory. The neoclassical growth (Solow-Swan) approach ignores macroeconomic stability as it is assumed that planned investment always equals planned savings, and that the capital-output ratio can change through investment. The macroeconomic adjustment process that makes this true is omitted. The Harrod-Domar approach has a fixed capital/output ratio and generally no equality of savings and investment, and results in unstable steady-state growth. Essential for growth and development in the Harrod-Domar framework, is the proportion of income saved (not consumed), which will steer capital investment - the engine of growth.

A central concept in neoclassical growth theory is convergence. The Solow model (Solow, 1956) has provided the basis for examining convergence. Absolute convergence means that different countries with the same population growth rate, the same savings propensity, access to the same technology but different capital-labor ratios, will converge to the same growth rate, steady-state capital-labour ratio, output per capita, and consumption per capita. Conditional convergence means that countries which differ in savings propensities and capital-labour ratio, will converge to the same growth rate but not necessarily to the same capital-labour ratio and output per capita. The lack of convergence has been explained theoretically by invoking notions of technological and population induced poverty traps (a vicious circle of low savings and few investment opportunities). In this sense the (Malthusian) demographic transition is also relevant: as income per capita rises, the population growth rate rises.

Exogenous growth theory does not explain empirical facts of growth well. In particular, about half of historical growth could not be attributed to investments in production factors. This so-called 'Solow residual' was one reason to develop endogenous growth theory (Barro and Salai-Martin, 1995; Aghion and Howitt, 1998). An essential assumption here is that private and public investments are associated with positive externalities (spill-overs) that undo tendencies of diminishing marginal returns on capital investment. Complementary investments in human capital (knowledge, innovation) and an optimal trade-off between human and physical capital investments are critical for this area of research.

Finally, two-sector models (with consumption and capital production sectors and intermediate deliveries) have received little and/or ad hoc attention. This is unfortunate, as they seem to be crucially related to transitions, in particular to understand structural changes in sector composition and interrelations. Endogenous growth theory has generated some two-sector models, but these are only meant to address the production and costs of innovations in a separate 'knowledge sector'. An original model is one in which the neoclassical rational agents and equilibrium are combined with vertical innovation or (Schumpeterian) creative destruction (Aghion and Howitt, 1992). All in all, the results of the enormous body of growth theory are a bit disappointing where transition research is concerned.

#### *Economic history*

We indicated in Section 1 that transition research would benefit from the study of historical transitions. This immediately supports the potentially useful role that economic history can play. In this field, all types of variables are interwoven into narratives (traditional approach) or combined in formal models (cliometrics). Economic history is much more data-driven than theory-driven. Theory is nevertheless implicit in the choice of explanatory variables and in the importance attached to certain variables and events. Mokyr (1990) has identified a long list of factors that he regards as crucial for successful technological innovations in economic history. Table 1 present a selection of the most important ones, which van den Heuvel and van den Bergh (2005) have divided into basic and side conditions. Although such a classification is helpful, it should be realized that the factors are not always independent. Factors may be strongly dependent on others in a certain region and during a specific time period. In any case, transition studies might systematically examine the state and role of these various factors.

	5
Basic conditions	Side conditions
Level of nutrition	Geographical environment
Life expectancy	Demographic factors
Norms and values	Institutions, property rights, and regulation
Openness to information	Resistance to innovation (conservatism)
Willingness to bear risks	Science and technology
Religion	
War	

Table 1. Fundamental innovation factors

*Note:* Factors based on Mokyr (1990); own classification in two types. *Source:* van den Heuvel and van den Bergh (2005).

Cliometrics, with its focus on quantitative analysis, might also be useful for understanding transitions. This type of study involves the application of economic theory and econometric-statistical techniques to understand regularities of economic history. Both Greif (1997) and North (1997) emphasize that this has advantages over narratives, even though a main limitation of the approach in the past has been associated with a strong reliance on foundations in neoclassical economics. North, however, is optimistic about using cliometrics to include such aspects as bounded rationality, transactions costs and changing preferences. In addition, a main problem to be resolved is associated with the linking of independent but often abundant data on economic, demographic, educational, institutional and technological variables. There is much to learn from past cliometric research, as major transitions have already received ample attention.

A famous example of this research includes Fogel's (1964) study of the transition from horse wagon to rail in the U.S.A.. Fogel argued that the common belief that railroads played an important role in overall American economic development at the end of the 19<sup>th</sup> century is overestimated. He suggested that the innovation of the railroad delayed the emergence of modern cars and trucks with combustion engines, for decades. In addition Fogel adopts a type of spatial economic perspective in assessing the impact of different (hypothetical) transport technologies based on spatial economic patterns.

An institutional-historical approach towards transitions is promoted by North (1981) in the analysis of the transition from primitive agriculture to modern intensive agriculture. North regards the move from common property (or better: open access) to exclusive property rights as the critical factor in the first economic revolution, because it allowed cumulative technological improvements that set the stage for a lasting period of high growth. Institutions may be composed of both formal (laws) and informal (social norms) rules of the game within a society (North, 1991). Modern economies require property rights, and effective, impersonal contract enforcement. Societies are less likely to develop and flourish when constraining formal or informal institutions limit the range of economic combinations possible. Unlike Rostow, North did not develop an economic model of a transition but focused on the role of political and economic institutions in reducing uncertainty and transaction costs in order to realize potential gains from exchange.

Another interesting historical study is provided by Wilkinson (1973), who developed an ecological explanation which linked the Industrial Revolution to natural resource factors (see also Common 1988). It recognizes a number of human strategies that respond to resource scarcity, such as new techniques, new resources, new goods, and migration. According to Wilkinson, the use of coal was stimulated by high prices of wood, following the significant loss of forest cover in England. In the early phase, coal was mined at strips at the surface. Later deep mines were explored. This in turn created an 'important' problem: the need to pump away groundwater, which gave rise to the first large scale, application of the steam engine. Widespread use of it gave rise to various refinements of the steam engine, alternative models and as such, new types of applications. In a next phase, spin-offs to other sectors occurred, especially to the textile industry and to transport's ships and trains powered by steam locomotives. Wilkinson's work helped to reconfirm the importance of resource scarcity for human development and advanced a cause-effect view which is very relevant to transition processes. The earlier model by Faber and Proops (in the previous section) can be regarded as a crude formalization of Wilkinson's conceptual model.

### Economic studies of transition economies

Since the 1990s much attention has been given to the transition from communism to marketdemocracy systems. This research was motivated by fundamental changes in the former USSR and east-European countries which have been taking place since the second half of the 1980s. Studies give specific attention to emerging markets (competition), institutions (property rights), transformation of public to private production, role of trust and independent media, entrepreneurship and investment, trade and foreign involvement. Economics of transition is now a well-developed field. In this field, there is unanimous agreement that institutions matter and that each country's evolution is unique.

There has been a fundamental debate between big bang and gradualist advocates, even though the knowledge basis is very thin. Economists have certainly underestimated the coordination problems of moving toward market-based systems (Gros and Steinherr, 2004). All European transition economies experienced a fall in output, for much longer times and more deep than originally anticipated. It required new models to explain this, with positive non-linear feedback (e.g., Rosser and Rosser, 2004). Countries that implemented reforms early on into the process usually were ahead of the others (Gros and Steinherr, 2004, p. 110). Reforms consisted of

price liberalization, privatization of property, liberalization of trade through removal of tariff and non-tariff restrictions, currency reforms, and macro-economic stabilization policies (to reduce inflation). Command-economy characteristics did, however, not disappear immediately or completely (path-dependency).

Within this genre of literature 'transition curves' are being estimated (Philips and Sul, 2005). Political economy aspects are investigated to explain differences in economic performance among transition economies in relation to the sequence of reforms (Roland, 2002; Kornai, 2005). The transition experience very much reinforced the institutional perspective in economics. It also stimulated people "to think about institutions not in a static way but in a dynamic way: how momentum for reform is created and how institutions can evolve, but also how momentum can be lost and how one can get stuck in inefficient institutions" (Roland, 2002, p. 47). The mainstream economics influence here is that governments should just improve the conditions under which decentralized decisions are made, i.e., let markets find the route towards the future (Hayek, 1978). All in all, much can be learned from this field for transition research.

#### Economics of technology

It is evident that the economics of innovation and technology address a number of issues that are relevant to transition research. This field is concerned with incentives and institutional mechanisms of knowledge creation and spillovers. Schumpeter is considered to be the founder of this field, at least where the definition of the issues is concerned. Ever since Schmookler (1966) innovation is seen as being responsive to economic stimuli with search processes being informed by engineering ideas that indicate what is worthwhile ('notional opportunities'). Later research on technological regimes found that innovation is not fully responsive to economic stimuli and occurs within certain patterns and constraints (Dosi, 1988; Malerba and Orsenigo, 1997). Innovation is found to be different across sectors and to be surprisingly similar across countries. This has to do with technological opportunities, appropriability conditions, degrees of cumulativeness of knowledge, and the nature of knowledge (generic versus specific, tacit versus codified, complex versus simple), as shown by Malerba and Orsenigo (1997).

Innovation is a very broad term, comprised of technological innovations, product innovations, and changes in markets, organisation, logistics and finance. Freeman and Perez (1988) make a distinction between incremental innovations, radical innovations, changes of technology system and pervasive changes in a nation's technological basis (changes in technoeconomic paradigm). Rosenberg talks about minor and major innovations where major innovations are those that provide a framework for a large number of subsequent innovations. For transition research, an important distinction is between sustaining innovations and disrupting innovations (Christensen, 1997). Disrupting innovations are those innovations that render obsolete existing structures and systems (e.g. automobiles and minicomputers). These innovations often come from outsiders who are serving non-mainstream customers. Disruptive products tend to under-perform in mainstream markets but have certain features that are highly valued by specific customers (i.e., military and other idiosyncratic costumers for who performance is more important than price). As a product improves, it might then break out of its original, small niche and replace the dominant product, as happened in the case of gas turbines and more recently in the case of digital cameras.

A transition model of technology evolution is the lifecycle model of Abernathy and Utterback (1978). The transition is the shift of product innovation, to process and unit production, to mass production, following the establishment of a dominant design. This model applies to high-volume products and does not apply to all goods. In services there may be a reverse lifecycle (Barras, 1986), with product innovation succeeding process innovations (automation). The lifecycle model has been extended into an evolutionary model of variation, selection and retention by Tushman and Rosenkopf (1992). They distinguish two phases: an era of ferment (a time of competing

designs and organizations around them), and an era of incremental change. During all stages there is variation, selection and retention.

The economics of innovation is a broad field drawing on information and behavioural economics (learning, bounded rationality), organizational theories and system theory. Attention is given to wider issues besides the economic stimuli, that are shaping technical change. The notion of a national system of innovation model is an effort to understand the whole of factors shaping processes of knowledge creation and dissemination. Special attention is paid to institutions of science and education, to firm capabilities and to cultural issues. Companies are seen as embedded in innovation systems (or regimes) that guide, aid and constrain the innovation. The task for policy is not so much to create more competition but to set in motion processes of cumulative causation (Jacobsson and Bergek, 2004). To do so, policy should strengthen inducement mechanisms and deal with blocking mechanisms. Many blocking mechanisms exist. Attention is given to the dominance of existing systems that have benefited from learning processes and processes of adjustment and integration (Freeman and Perez, 1988, Kemp and Soete, 1992, Freeman and Louca, 2001).

A formal analysis of technological succession is being offered by Windrum and Birchenhall (2005). Increasing returns to scale are found to play an important role in successions, fitting in with empirical findings (for example of Klepper, 1996) that high start up costs are an important factor deterring new market entrants. A succession, a specific kind of transition, is found to be more than the displacement of old technology products by new technology products. It also involves the displacement of existing customer classes and preferences, the displacement of established market firms and established structures of production.

#### *Environmental, resource and ecological economics*

A transition to sustainable development cannot occur without good environmental regulation and resource management. The field of economics dealing with this is environmental (and resource) economics. Environmental economics addresses the economic analysis of the causes and the nature of environmental problems and their solutions. This includes issues relating to markets as well as to public policy. Environmental economics covers resource economics. The combination makes sense, since many resource issues are intricately linked to environmental issues. This is perhaps most noticeable in the case of fossil energy resources, the use of which contributes to the enhanced greenhouse effect. The notion of sustainable development, which has become one of the pillars of modern environmental economics, supports the linkage of resource and environmental problems. Closely related to environmental economics is ecological economics. It can be regarded as a social-science oriented version of environmental science. It has perhaps been most successful in promoting multidisciplinary research in which natural scientists (notably ecologists) and social scientists (notably economists) join forces.

The economic theory of environmental policy starts from the concept of "externalities" (see also the item "public economics"). Environmental economics is particularly interested in negative environmental externalities, i.e. negative physical effects of environmental pollution, resource use, or other types of environmental disturbance, such as fragmentation due to road infrastructure in nature areas, by one agent to another. Externalities have been analytically examined and discussed with the help of partial and general equilibrium theories, which is consistent with neoclassical assumptions regarding individual behaviour and operation of markets.

Instruments of environmental policy are traditionally evaluated in economics on the basis of their efficiency features. Effectiveness and distribution effects (equity, fairness) function as secondary evaluation criteria. The most common (archetypal) comparison is between uniform standards and taxes of pollution control. Taxes are attractive as they provide better incentives than standards to change individuals' behaviour, and thus realize more efficient outcomes: either social welfare is higher or costs of realizing fixed targets are lower. This is accomplished by equalising marginal costs of pollution abatement, assuming that individual polluters are minimizing costs. Standards are especially attractive from the perspective of effectiveness (or uncertainty). A combined instrument is a system of tradable permits. This has two features - a ceiling is set on all pollution emissions of a particular type by granting a finite amount of emission permits and, permits are tradable. The first feature assures effectiveness, and the second efficiency. Taxes and tradable permits are dynamically efficient, as they provide a permanent incentive for cost-savings through innovations.

Environmental economists support transition research by studying the irreversibility of changes resulting from policy and/or a lack of policy. This has been addressed with option value theory (Arrow and Fisher, 1974). Decisions made in private markets do not lead to socially desirable use, allocation, transformation and conservation of nature and (scarce) land. This literature suggests that under certain conditions irreversible developments should be avoided or postponed until better information can be obtained. See Gollier et al. (2000) for linking this literature to the precautionary principle, and Pindyck (2000) for examining the notion of optimal timing for implementation of environmental policy.

An important and somewhat related idea, receiving much attention in ecological economics, is 'resilience', an extended stability concept. It has two definitions: (i) the time necessary for a disturbed system to return to its original state (Pimm, 1984); and (ii) the amount of disturbance that a system can absorb before moving to another state (Holling, 1973, 1986). Unlike most growth economists, whose models provide no upper bound on economic growth, physical scientists and ecologists are accustomed to the idea of limits. Ecologists are far less optimistic about substitution possibilities (of man-made capital for natural capital) than neoclassical economists, which is evident in the Holling view of sustainability which is really about maintaining ecosystem resilience. Diversity is important for resilience and for evolution.

Some authors have tried to find analogies of resilience in socio-economic systems. Static efficiency achieved through standardisation or tight control may reduce resilience (adaptivity) through reduced diversity. (see Levin et al. 1998; Gunderson and Holling, 2002). Policy implications are very different from those of standard economics: diversity is to be positively valued, one should be careful not to overexploit, and one should not exceed critical thresholds leading the system into collapse. This provides a bridge to evolutionary approaches which might shed light on the particular micro-level mechanisms involved in bifurcation and resilience (Rammel and van den Bergh, 2003).

Other ideas in ecological economics with relevance to transition research are coevolution and endogenous preferences, both having close connections to evolutionary thinking. Norgaard (1984) was the first to consider the application of the concept of coevolution to the interaction between economic and natural systems. His main illustration is the coevolution of pests, pesticide and environmental policy in the U.S.A. (e.g., Norgaard, 1994). Nevertheless, his ideas have been criticized for not indicating a sharp distinction between the interaction among subsystems ("codynamics") and strict coevolution as interacting populations with internal diversity causing mutual selection pressure (Winder et al., 2005).

The idea that preferences are endogenous instead of invariant has led Norton et al. (1998) to argue that changing consumer preferences can be in and of itself, an instrument of environmental policy. In particular they state that stable preferences are at best realistic over short periods of time, and that sovereign preferences are inconsistent with long term goals of sustainability. Consequently, public discussion about ethical consumption and sustainability should be stimulated via education, advertising rules, cultural norms, etc. Changing consumers' preferences through democratic processes could be used to encourage environmentally conscious consumption in a way that consumers would not feel "... deprived and unhappy ..." but "... enlightened and happy after being educated into the joys ..."(Norton et al., 1998, p203). Most democratically elected governments have already formulated public policies aimed at influencing norms which are regarded as criminal, racist or otherwise undemocratic, so why not extend this

practice towards environmental sustainability? A demand side perspective is a useful addition to the dominance of innovation and R&D supply-side oriented policies, necessary to realize complex transitions.

#### Industrial economics

It is important to understand the behaviour of firms, their responses to incentives and regulation, and the interaction between firms in markets when trying to stimulate transition processes. Industrial economics offers relevant insights, being an extension of the theory of the firm to address a number of firm-internal and firm-external issues in a theoretical manner (e.g., Martin, 2001). This includes attention to market type (monopoly, oligopoly, etc.), vertical integration, price strategies (multi-product pricing, intertemporal price discrimination), product differentiation, price-quality trade-off and advertising, strategic behaviour and interaction between firms, the role of capacity constraints and investment strategies, rent seeking, complementary and networks, and asymmetric information affecting management and competition. For transition research especially long term issues like long-run, sunk and transactions costs, long (vs short) run price competition, barriers to entry and entry deterrence, exit, market contestability (threat of competition for incumbent firms by potential entrants), and R&D strategies are relevant when conducting transitions research. The policy angle within industrial economics focuses on antitrust legislation and other regulation to overcome various market failures related to imperfect markets (notably, collusion), incomplete and asymmetric information, standards and lock-in, and innovation externalities (patents, subsidies). A more empirical and statistical branch addresses the demography of industries (Carroll and Hannan, 2000).

### Path-dependence and lock-in

A specific concept on the boundary of economics of technology, evolutionary economics and industrial economics, is path-dependence. This concept is highly relevant for transition research, as one can regard the essence of the transition problem to be the un-locking of a locked-in socioeconomic-technological-institutional system. Path-dependence is intricately linked to lock-in. Moreover, lock-in has been one reason, additional to already recognized 'market failures', for believing that markets do not generally lead to (socially) efficient, or otherwise desirable, outcomes. Whilst markets encourage efficiency in production, the long-term outcomes may not be the most efficient when demand and supply are characterised by increasing returns. Regulation of environmental externalities may not have the intended effect of moving to an alternative technology or product associated with less environmental pressure. A technology that has achieved an early advantage may dominate or capture the market due to increasing returns to scale, according to a kind of self-reinforcement process, i.e., a larger market or market share stimulates relatively high growth. Increasing returns to scale take a number of forms:

- (Traditional) scale economies: cost and price decrease as the scale of production increases.
- Learning by using or doing: improvement, lower costs (producer), better performance (consumer).
- Imitation or bandwagon effect: rather than innovate, companies may imitate the most successful product, and the popularity of a certain good may propagate (as with fads).
- Agglomeration effects: spatial spillovers, easy communication and short transport distances that give rise to positive externalities.
- Network externalities: being connected to a larger network (e.g. via phone) often has an advantage.
- Financial power: the more a product or technology is adopted, the more resources will be available for its development and perfection, which, in turn, will make it relatively attractive for potential adopters.

- Informational increasing returns: if a product is adopted more, and therefore becomes better known, then risk-averse individuals will more easily be convinced to buy it.
- Technological inter-relatedness or complementarity: infrastructure and sub-technologies are often complementary (gasoline, refineries, filling stations, car technology). This relates to the importance of coevolution of complementary factors (knowledge, technology and institutions).
- New knowledge is shared by firms: firms increase knowledge due to learning-by-doing and innovation, which can be shared with other firms in the same sector with similar problems. This leads to dynamic increasing returns at the level of industries.
- The political power of vested interests, operating, among others, through lobbying in politics.

Due to these mechanisms, formally studied by Arthur (1988, 1990), a technology with a large market share, due to an early start, has an advantage and as such can grow relatively quickly (attain an even larger market share), even without any intrinsic (cost or net welfare) advantages. An important consequence of increasing returns is that the (adoption) process towards the final or equilibrium state of the system is path-dependent (non-ergodic). Path-dependence can be interpreted as temporally remote events having a significant or dominant impact on the present: the logic of the world can be understood only by uncovering how it got this way (David, 1985). Path-dependence thus implies irreversibility. It should be noted, however, that path-dependence is not the consequence of purely random factors, but results from interaction between random and systematic or deterministic, notably selection, factors. Contingency interacts with historical selection forces, with the latter probably being more important. In the words of historian Landes: "big processes call for big causes". This fits with the transition perspective of Rotmans and others who assert that mutually sustaining developments at different levels (niches, regimes and landscape) are necessary for a transition to occur.

Path-dependence can result in lock-in. Lock-in occurs when a dynamic pattern of competing technologies ends up in a situation with one technology dominating the market. Heterogeneity of user needs and other types of heterogeneity (at the supply side or otherwise) usually mitigate against this but, in the case of large increasing returns, as with network technologies, there is a danger of becoming locked into particular solutions. A situation of lock-in can persist for a very long time, as happened with the QWERTY typewriter (David, 1985) and the light-water nuclear reactor for electricity production (Cowan, 1990). It can be argued that the use of fossil fuels constitutes an example of lock-in. The latter is relevant to the transition to a sustainable economy and energy system.

Apart from the economic sources of lock-in we have institutional and cultural pathdependency, meaning that institutions or cultural habits are fixed and cannot be changed easily (Unruh, 2000, 2002; Kemp, 1994). Any transition requires new institutions and new forms of alignment, which take time and effort. They are not controlled from a central point but the result of distributed agency and entrepreneurship. In the establishment of a new path, a multitude of actors becomes involved with a technology and generates inputs that result in a transformation of an emerging technological path (Garud and Karnoe, 2003). In the early period when the technology is in a flux, social, organizational and political factors are important shapers of technical change, later on 'technology' becomes a determining factor, with technological imperatives setting the conditions for competition (Tushman and Rosenkopf, 1992; Molina, 1999). In conclusion, it can be said that any new path creates new path-dependences.

#### Public economics

Public economics is the sub-discipline within economics that provides the most general perspective on public policy, in terms of the conditions under which public intervention is required as well as the form policy should take (institutional arrangement, policy instrument

type). Likewise, public economics can shed light on certain aspects of transition management, notably where these relate to solving public goods and externality type of problems. Pure public goods are characterized by non-rivalry and non-excludability in use. However, many mixed type of private-public goods are found, namely where rivalry is not perfect due to congestion beyond a certain threshold, or where property rights take different forms (common, state, clubs). To make things more complicated, both rivalry and excludability may sometimes change due to technical progress (e.g., technical exclusion or inclusion, e.g., allowing the whole world to see a particular football match).

A number of public goods are relevant in the context of transitions. One is knowledge, which is a quasi-public good. It is non-rivalry in use but people can be excluded from it, through secrecy and patents. Much public investment in knowledge is undertaken most notably through universities and related institutes. Certain goods require physical infrastructure for their use. Historically government plays an important role in the provision of such infrastructure but it can also be provided through private capital if the costs of using it can be charged to the users (e.g., toll roads). Public-private-partnerships offer a mechanism to transfer the risk and debt associated with these new capital projects to the private sector in return for a long-term service contract. Property rights typically remain with government.

Externalities are unintended and unpriced (outside markets) effects exerted by one economic agent onto another, affecting his/her utility or profit. Externalities come in both negative (e.g., environmental pollution) and positive forms (R&D or knowledge benefits). The fact that individuals make decisions without concern for externalities – by definition – means an outcome that is not consistent with the highest attainable level of social welfare. Regulation is therefore required. Negative externalities need to be suppressed (optimally with good taxes, levies or tradable permits) while positive externalities need to be responded to with subsidies or patents.

Next to policy, self-regulation or evolution of social norms of common-pool resources can be an effective mechanism (Ostrom, 1990). This can occur through local interactions among users that involve monitoring and (altruistic) punishment. Externally or hierarchically imposed regulation can destabilize co-operation, notably when associated hierarchical monitoring is imperfect. Stimulating norms through communication may then be more desirable. Instability in the evolutionary equilibrium can also arise when certain parameter or external changes occur. Examples are: sanctions decline, harvesting becomes more efficient (technical progress), the resource price alters, resource users migrate, external economic and political events, and natural disasters influence the resource quality and processes.

Finally, public choice theory clarifies how public decision making and policies are influenced by special interests through lobbying and negotiation. In addition, it recognizes that a government is not a homogeneous group or analogous to a single decision making individual, but is composed of various individuals or groups (civil servants and politicians) with possibly conflicting interests and goals (Dietz and Vollebergh, 1999). This is of relevance to transition research, where it is often stressed that important barriers to transitions are created by power and vested interests of specific stakeholder groups.

#### Spatial economics

Transitions involve heterogeneous spatial processes as well as complex spatial interactions. Spatial issues are studied within spatial economics, which is an amalgam of regional, urban and transport economics. Spatial economics cover issues that use the approach of neoclassical economics and approaches common to the study of technological innovation and spatial diffusion, to understand spatial economic regularities. Important insights of this field for transition research relate to the location of industries, urban and regional markets, spatial externalities (both negative and positive - innovation and agglomeration related), spatial patterns of diffusion, regional diversity, regional isolation, regional (open system) dynamics, and

economic effects of infrastructure. Recent development in new economic geography furthermore link up with international economics (Krugman, 1991).

One only needs to refer to the problems associated with non-point source (diffuse) pollution problems, land use and biodiversity loss, specific urban environmental problems (noise, health problems, congestion), and the impact of stringent environmental regulation on location choice, trade and transport (Siebert, 1985; van den Bergh, 1999, part V), to understand that the interaction between environmental policy and spatial economic analysis is relevant.

The interaction between technological innovation studies and spatial economic analysis has focused on regional development, spatial diffusion of innovations, and diversity and isolation of regional initiatives (Malecki, 1997; Acs, 2003). Insights about the functioning and boundaries of urban and regional markets – for products, services, housing, labour, and transport – may moreover be useful when searching for niches and diverse industrial trajectories. Agglomeration effects may be important as well, similar to the role played by positive externalities in endogenous growth theory. Fostering agglomeration means stimulating shared inputs (labour pool, knowledge) at relatively low costs. In addition, public infrastructure and capital may be crucial (i.e., roads, public transport, parking facilities, communication systems, and attractive nearby living conditions). An important question with regard to public investment in infrastructure is whether it stimulates new development and innovations or just leads to relocation of existing activities (Eberts and McMillen, 1999).

From a policy perspective, a number of issues are relevant. Diffuse pollution might require input regulation rather than direct pollution regulation. The Tiebout hypothesis is relevant (Tiebout, 1956) with respect to local public goods. This hypothesis points out a mechanism of self-organization at a higher spatial level by suggesting that local public goods may differ and in turn cause migration ('voting with the feet'). Implementing land use taxes will affect the cost of using land, which in turn will influence location choices by firms and households. Henry George suggested already in the 19th century to tax the rental value of land. This will generate revenue that benefits the community instead of the individual private landowner. Very often the price of land is high simply because of positive externalities generated in the surrounding area by attractive places and services (Cohen and Coughlin, 2005). Likely consequences of a single (Georgian) or two-rate land tax are an improved quality of land and buildings, shorter transport distances, and less urban pollution. Such land taxes can possibly contribute to a transition to sustainability.

#### 4. A summary of contributions from economics

Given the broad scope of this article it is impossible to provide a complete overview. Nevertheless, it is worthwhile to try to identify the most exciting theories and concepts that economics has to offer for transition research. Table 2 outlines the results of our efforts. Its point of entry indicates the type of problems that one typically experiences while studying transitions, whether reflecting into the past (history) or visually the way ahead (future). The table summarizes the main approaches and concepts by identifying a list of core transition challenges in the first column, and then relating these to economic approaches. Relevant concepts of these approaches are listed and where necessary, comments are provided to indicate if there is a requirement to elaborate further on the approach or concept(s) as they relate to the respective transition challenges, which in reality are often connected and interactive. Similarly, sub-disciplines of economics are not entirely independent, but overlap in terms of problems studied, concepts used and solutions offered. That said, the table serves a useful function by providing guidelines for the reader interested in transitions research and the schools of economics that seek solutions to the problems specific to transitions.

Tuble 2. Int Overviev	v oj ceononiic incortes ana concept	is relevant to transition resear	
Transition challenge	Useful economic approaches	Relevant concepts	Need for elaboration
Barriers	Grand development theories, Economic history, Evolutionary economics, Ecological Economics	Resilience, lock-in, learning by using or doing, imitation, network externalities, technological inter- relatedness (complements), vested interest (rent-seeking)	+
Supply side	Economics of technology, Industrial economics, Grand development theories	Market failures, scale economies, innovation strategies, cooperation, networks, complementary products, vertical integration, rent seeking, demography of industries	0
Transition dynamics	Long waves, Economic history, Macroeconomics, Growth theory, neo- Austrian economics, Development economics, Economics of disasters Economics of technology, Evolutionary economics, Economics of information	Increasing returns, life cycle, competition, multi-stages, shocks, disasters (natural, economic, war)	++
Innovation tempo		Patents, subsidies, market niches, experiments, diffusion, adoption, innovation systems	+
Demand side	Marketing, Behavioural economics	Habits, imitation, lexicographic preferences	++
Externalities	Public economics, Environmental economics	Public goods, club goods, regulatory instruments, efficiency	0
Diversity	Evolutionary economics, Economics of technology, Economics of information	Short term efficiency, long term innovation, increasing returns, adaptation, flexibility, real options	+++
Spatial organization	Regional and urban economics	Location, Tiebout hypothesis, agglomeration effects, transport, infrastructure, regional diversity, spatial isolation, trade	++
Institutional conditions	Institutional economics, Public economics, Economics of transition economies	(Intellectual) property rights, education, science, legislation, markets, culture and behaviour (norms and values, risk attitude, conservatism, religion), regulation	++
Knowledge	Economics of information, (Endogenous) Growth theory	Asymmetric information, learning, search, uncertainty, expectation, positive externalities, network formation, informational increasing returns	+
Role of private firms	Industrial economics, economics of information	Competition, market failures, investment conditions, uncertainty	0
Role of government(s)	Public economics, Environmental economics	Government failures, market (self- organization), public investment (infrastructure, R&D), removing uncertainty (insurance), conflict resolution	++
Demography	Population economics	Education, labour force, productivity, risk attitude	+

Table 2. An overview of economic theories and concepts relevant to transition research

#### 5. Evaluating mainstream economics from a transition research perspective

So far we have looked rather constructively at the theories, concepts and insights economics broadly offers a potential contribution transition research. It should be acknowledged, however, that some authors have emphasized that the dominance of mainstream or neoclassical economics provides a barrier to understanding and perhaps even fostering urgently required transitions, and that it conflicts with the 'transition (management) paradigm' (Rotmans, 2003). A problem in assessing the value of the latter statement is that this 'paradigm' has not yet crystallised out (RMNO, 2003). Indeed, it is still characterised by a diversity of opinions and a loose set of ideas without firm and consistent theory and associated formal models. In fact, this incomplete development of transition theory underlines the usefulness of examining what the contribution of economics might be to understanding and managing transitions.

The question therefore is this: how well suited is neoclassical economics for the study of long term transitions. A quick judgement might be that neoclassical economics is too concerned with economic variables, prices and income effects, and is missing out on the cultural, institutional and structural aspects of economic change. In defence of neoclassical economics, apart from what was already noted in section 3.1, we can say that especially general equilibrium models provide a complete structure of how prices and quantities of all the goods are linked together. A transition involves large changes, with effects that permeate the economy (among others) through price and cost structures. In this respect, the neoclassical economic approach provides a firm tool for calculating the impacts of structural change. Economists prefer to separate out only a few cultural and institutional aspects but this does not mean that they are not part of the equation: they are implicit in the demand and cost specifications. The economist's way of viewing the world is in terms of markets (demand and supply interaction) for labour, capital, other input factors, and goods and services. These various kinds of markets interlock, which is described in the most complete manner in general equilibrium models. Markets solve complex coordination problems of distributed knowledge by signalling relative scarcity thereby guiding the plans of economic actors at the supply and demand side. This is what economists are preoccupied with. Neoclassical economics cannot provide the complete picture of transitions. But this is hardly a serious criticism. No single theory can cover the entire spectrum and complexity of long term transitions. .

Nevertheless, one can wonder whether dominant approaches in economics include the right variables and processes, and employ the correct behavioural assumptions, to study long-term transitions. The relevant processes will differ from case to case, as is already clear from the typology of transitions by Geels and Schot (2005). Economic studies may especially be criticised for giving little attention to changing institutions, which often will imply altering model parameters or structure. In order to identify and analyse the various transition processes as well as incorporate institutional dimensions, collaboration between economists, sociologists, business historians and other social scientists might be useful.

In addition, non-linearities and discontinuities play a role in transitions. Examples are the creation of new products and the emergence of institutions through agent interactions. Traditional equilibrium models with representative agents tend to underestimate such interaction effects. In this respect, agent-based models are more flexible and powerful. They allow for addressing gradual changes in the degree of interaction (or coordination) or gradual changes in behaviour, which in turn can lead to discontinuous changes (Rosser, 1999). Moreover, whereas the microfoundations in mainstream economics emphasise upward causation, a more realistic approach might be a combination of upward and downward causation (van den Bergh and Gowdy, 2003). The latter seems to be consistent with the multilevel transition framework. It provides an opportunity to incorporate difficult issues relating to emergent phenomena, such as network and group formation and, appearance of new norms.

Coevolution theories appear useful when attempting to understand the complexity of transitions. The transitions studies by Geels (2005) show that the relations between niches and regimes can be complex: existing regimes can make new niche developments difficult through competition with the established regime; niche development can, however, occur within a dominant regime (e.g., gas turbines were used within a coal regime before they overtook it); and niches outside a regime may benefit from certain regime features (e.g., the use of roads by non-motorized transport).

Finally, economics has been criticized to be focused on efficiency or cost-effectiveness, cost-benefit analysis, and optimality and smooth changes, in contrast with transitions that involve uncertainty and discrete jumps. This view, however, oversimplifies matters. Even though in practice many people and firms focus on strategies that reflect short term efficiency (e.g., more attention for energy conservation than renewable energy), economic theory does not imply this. Neoclassical economics does not exclude a search for solutions that satisfy long term efficiency (e.g., possibly renewable energy). It is true, perhaps, that efficiency is often approached in too deterministic a manner, giving too little attention to uncertainty and irreversibility. On the other hand, advanced approaches to evaluation in economics, such as option value and real options theories, do address uncertainty explicitly (Dixit and Pindyck, 1994; Fisher, 2000).

Perhaps more important is that neoclassical economics assumes rational agents that show optimal behaviour. According to an evolutionary perspective, dynamic efficiency through new economic development may evolve from initially sub-optimal behaviour, namely from actors doing something different. As Peter Allen has said: "In an evolutionary landscape of hills and valleys representing levels of functional efficiency of different possible organisms, it is the error-maker who can move up a hill, eventually out-competing a perfectly reproducing rival. And this despite the fact that at each and every instant it would be better not to make errors, since the majority are loss-making" (Allen, 1988, p. 107). Diversity is to be positively valued. The yardstick of (short-term) efficiency for policy action may not always be a good guide in an evolutionary world. Short term efficiency and long term efficiency may conflict, and diversity is in between. Neoclassical economics, however, does not attach value to diversity, but rather sees it as a cost. Incorporating diversity and the related themes of path-dependency (seen as irreversible change of diversity) and lock-in (seen as minimal diversity) into mainstream economics could mean an important contribution to transition research.

Finally, economics has stressed that there are several market failures which cause a deviation between market outcomes and social welfare optima. These market failures include:

- Missing markets: no markets for all goods and services.
- Imperfect competition on some markets.
- Economic activities generating externalities.
- The existence of public goods.
- Property rights being incompletely assigned.
- Transactions not occurring under perfect information.
- Bounded rationality: cognitive limitations and routines preventing firms from maximising profits and individuals from maximising their utility.
- The presence of transaction costs.

These various market failures have clear responses in terms of governmental policy or regulation. Economics offers concrete suggestions. In as far as transitions are hampered by such failures, economics can thus offer solutions. The most important lacking (system) failure seems to be path-dependence and lock-in, but even in this case economists have been able to convert it into the neoclassical paradigm, namely as increasing returns to scale, i.e., a kind of positive externality (Gerlagh and Hofkes, 2002).

#### 6. Conclusions

The broad discipline of economics with its various sub-disciplines has much to offer for transition research. We have provided a broad overview here, without trying to create any personal selection bias, but instead allowing the reader to identify the concepts and theories of interest to facilitate stream-lined access to the specific economic literature that is relevant to certain aspects of transitions research. A disadvantage of our approach is of course that it skims the literature, and as a result is abstract and brief in its account of concepts and insights. This approach is intentional. Only a book format might overcome this shortcoming.

It may come as a surprise to non-economists involved in transition research that economists in the past have already generated many ideas that relate to transitions. For example, they have elaborated in great detail the notion of multi-stage development (notably Rostow), which seems to be reiterated in recent multi-stage theories. Economics can make many useful contributions to the study of transition. Perhaps the greatest contribution economics can make to transition research concerns the role of markets and limitations of planning in transitions, and the choice of instruments of environmental regulation. For dealing with public 'bads', such as pollution, economists favour market-based instruments such as pollution taxes and emission trading systems. They help to deal with them in a decentralized manner by not prescribing choice, which helps to achieve total reductions in pollution in a low-cost manner. Institutional economics can help make visible undesirable incentive structures and propose alternatives. Spatial (regional, urban and transport) economics provides many insights about the geographical and multilevel nature of economic activities and mechanisms. Path dependency may be analysed through evolutionary-economic models that allow for the understanding of the processes of lock-in at the micro-level. In addition, economic models might be elaborated to address multi-level aspects of transitions, including processes at the level of niches, regimes and landscapes. Diversity might be given a place in economic analyses to move from short term to long term efficiency, thus allowing a trade-off of short-term costs of diversity (keeping options open) with long term benefits (because diversity generates new options in the future). Bounded rationality (habits, routines, imitation, rivalry, group norms) may receive more attention in economic analyses, as this often provides a more realistic depiction of the response of individuals to complexity and uncertainty, which is relevant in certain transition phases.

We have briefly addressed the potential friction between the dominant neoclassical school in economics and the aims of transition research. Some writers in transition research have noted this potential friction. Our position is that certainly neoclassical economics does not offer the complete perspective on transitions that a concrete, formal theory does. Neoclassical economics offers useful information on market processes and economic motives, but tends to neglect or treat implicitly cultural and certain institutional aspects of transitions. It can provide insight into the how transitions will indirectly affect the economy through interacting markets, prices, costs, sectors, and demand and supply, with implications for labour, capital and other input factors. This appears to be a promising area for transition research, in which models of transitions to a market economy can play a useful role. As to the issue of achieving welfare benefits through system innovation, some critics have suggested that the economist's focus on efficiency or cost-effectiveness, using the tool of cost-benefit analysis, which prevents the search for uncertain, long term and discrete or non-smooth solutions and changes, and which means that there is no specific role for transition management. We feel that this oversimplifies matters. Economic theory does not prescribe short-term efficiency or optimisation. These are outcomes of the myopia of people, firms and governments that want to reap the benefits of investments as soon as possible. This exemplifies itself, for example, in the greater concern for energy conservation than for renewable energy. Perhaps efficiency is often approached in a too deterministic way, but on the other hand economics has developed advanced formal theories and models to deal with uncertainty and irreversibility, notably option value and real options theories. Perhaps what needs to be done is to extend these approaches to include the potential, future value of diversity, so that ultimately a good trade-off between diversity and short run efficiency is feasible. Last but not least, economics has recognized the problems associated with path-dependence and lock-in. They are in fact considered one of many market failures, namely as a positive externality. A third possible critique on mainstream economics is the assumption of perfectly rational agents giving rise to optimal choices and social outcomes (in theory). Here perhaps lies the weakest element of mainstream economics, which is currently under attack from the fields of behavioural, experimental and evolutionary economics, all of which offer useful alternative ways to approach market and strategic interaction and public regulation, for transition research.

Economics is concerned with optimality but is generally critical about planning as a way to achieve this. Governments should just improve the conditions under which decentralized decisions are made. The main contribution of mainstream economics is then perhaps that it offers specific insights about how to respond to a number of market failures that hamper transitions. Economists and technology researchers and historians seem to agree that governments cannot and therefore should not try to pick winners. It is better to rely on the (evolving) self-organisation capacity of institutions and markets, even when the operation of markets may occasionally lead to sub-optimal solutions through path-dependence resulting in a lock-in. Evolutionary economists would next to selection by markets see an important government role for stimulating or even contributing (through public R&D) to a variety of options, precisely because markets are myopic and guided by increasing returns to scale, thus running the risk of fostering (too) early lock-in.

Given the rich set of concepts, ideas and approaches identified here, economics without any doubt can play a useful role in transition research. Of particular interest are the subdisciplines of business cycle theory (long waves), development economics, economic history, economics of transition economies (former communist countries), economics of technology, environmental economics, institutional economics, public economics, and spatial economics. In view of this, a fruitful strategy would be to transfer relevant insights from each of these areas to transition research, and where relevant elaborate and synthesise them. This strategy would ideally involve interaction with experts from each of the sub-disciplines.

#### References

Acs, Z.J. (2003). Innovation and the Growth of Cities. Edward Elgar, Cheltenham.

- Aghion, P., and P. Howitt (1992). A model of growth through creative destruction. *Econometrica* 60(2): 323-351.
- Aghion P., and P. Howitt (1998). Endogenous Growth Theory. MIT Press, Cambridge, MA.
- Allen, P. (1988). Evolution, Innovation and Economics, in G. Dosi, C. Freemand, R. Nelson, G. Silverberg and L. Soete *Technical Change and Economic Theory*, Pinter Publishers, London, pp. 95-119.
- Arrow, K.J., and A.C. Fisher (1974). Environmental preservation, uncertainty, and irreversibility. *Quarterly Journal of Economics* 88(2): 312–319.
- Arthur, B. (1989). Competing technologies, increasing returns, and lock-in by historical events. *Economic Journal* 99: 116-131.
- Barras, R. (1986). Towards a theory of innovation in services. Research Policy 15: 161-173.

Barro, R. J. and X. Sala-i-Martin (1995). Economic Growth. McGraw-Hill, New York.

- Baumol, W.J. and W.E. Oates (1988), *The Theory of Environmental Policy*. 2nd ed. Cambridge University Press, Cambridge, UK.
- Bergh, J.C.J.M. van den (2001). Ecological economics: themes, approaches, and differences with Environmental Economics. *Regional Environmental Change* 3(1): 13-23.
- Bergh, J.C.J.M. van den, and J.M. Gowdy (2003). The microfoundations of macroeconomics: an evolutionary perspective. *Cambridge Journal of Economics* 27(1): 65-84.
- Bergh, J.C.J.M. van den, A. Ferrer-i-Carbonell en G. Munda, (2000). Alternative models of individual behaviour and implications for environmental policy. *Ecological Economics* 32(1): 43-61.
- Bergstrom, T.C. (2002). Evolution of social behavior: individual and group selection. *Journal of Economic Perspectives* 16(2): 67-88.
- Bleischwitz R. (2004). Governance of sustainable development: co-evolution of corporate and political strategies. *International journal of sustainable development* 7(1):27-43.
- Board on Sustainable Development (2000). *Our Common Future. The Transition to Sustainability.* US National Academy of Sciences, Washington D.C.
- Bruyn, S.M. de, and R.J. Heintz (1999). The environmental Kuznets curve hypothesis. In: J.C.J.M. van den Bergh (ed.). *Handbook of Environmental and Resource Economics*. Edward Elgar, Cheltenham.
- Butter, F.A.G. den, and M.W. Hofkes (2004). Technological transitions: a neo-classical viewpoint. In: A.J. Wieczorek and X. Olsthoorn (eds.). *Transformation Science*. Kluwer Academic Publishers, Dordrecht.
- Caldwell, B. (1984). Appraisal and Criticism in Economics: A Book of Readings. Allen and Unwin, Boston.
- Camerer, C.F., G. Loewenstein and R. Rabin (eds.) (2003). *Advances in Behavioral Economics*. Princeton University Press, Princeton, NJ.
- Carroll, G.R., and M.T. Hannan (2000). *The Demography of Corporations and Industries*. Princeton University Press, Princeton, NJ.
- Christensen, C.M. (1997). The Innovator's Dilemma, Harvard Business School Press, Harvard.
- Cohen, J.P., and C.C. Coughlin (2005). An introduction to two-rate taxation of land and buildings. *Federal Reserve Bank of St Louis Review* May/June 2005: 359-374.
- Common, M. (1988). Poverty and progress revisited. In: D. Collard, D.W. Pearce and D. Ulph (eds.). *Economics, Growth and Sustainable Environments*. St. Martin's Press, New York, pp. 15-39.
- Common, M., and C. Perrings (1992). Towards an ecological economics of sustainability. *Ecological Economics* 6: 7-34.
- Conlisk, J. (1996). Why bounded rationality? Journal of Economic Literature 34: 669-700.

- Cowan, R. (1990). Nuclear power reactor: A study in technological lock-in. *Journal of Economic History* 50: 541-567.
- Daly, H.E. (1991). Steady-State Economics. 2<sup>nd</sup> ed. Island Press, Washington D.C.
- Dietz, F.J., and H.R.J. Vollebergh (1999). Explaining instrument choice in environmental. Policies. In: J.C.J.M. van den Bergh (ed.). *Handbook of Environmental and Resource Economics*. Edward Elgar, Cheltenham.
- Dixit, A.K., and R.S. Pindyck (1994). *Investment under Uncertainty*. Princeton University Press, Princeton, N.J.
- Dorfman, R., P.A. Samuelson, and R.M. Solow (1958). *Linear Programming and Economic Analysis*. McGraw-Hill, New York.
- Dosi, G., Pavitt, K. and Soete, L. (1990). *The Economics of Technical Change and International Trade*. New York, New York, USA: Harvester Wheatsheaf.
- Durham, W.H. (1991). *Coevolution: Genes, Culture and Human Diversity*. Stanford University Press, Stanford.
- Eberts, R.W., and D.P. McMillen (1999), Agglomeration economies and urban public infrastructure. In: P. Cheshire and E.S. Mills (eds.), *Handbook of Regional and Urban Economics*, Vol.3. North-Holland/Elsevier, Amsterdam.
- Faber, M., and J.L.R. Proops (1990). Evolution, Time, Production and the Environment. Springer- Verlag, Heidelberg.
- Fagerberg, J. (1994). Technology and international differences in growth Rates. *Journal of Economic Literature* 32(3): 1147-1175.
- Fagerberg, J. and B. Verspagen (2002). Technology-gaps, innovation-diffusion and transformation: an evolutionary approach. *Research Policy* 31: 1291-1304.
- Fehr, E., and U. Fischbacher (2002). Why social preferences matter the impact of non-selfish motives on competition, cooperation and incentives. *Economic Journal* 112: C1-C33.
- Fisher, A.C. (2000). Investment under uncertainty and option value in environmental economics. *Resource and Energy Economics* 22(3): 197-204.
- Fogel, R.W. (1964). Railroads and American Economic Growth: Essays in Econometric History. Johns Hopkins Press, Baltimore.
- Freeman, C. (ed.) (1996). *Long Wave Theory*. The International Library of Critical Writings in Economics Series. Edward Elgar, Cheltenham.
- Gaddis, J.L. (2002), *The Landscape of History. How Historians Map the Past*, Oxford University Press, Oxford.
- Galor, O., and O. Moav (2002). Natural selection and the origin of economic growth. *Quarterly Journal of Economics* 117(4): 1133-1191.
- Geels F.W. (2002). Technological transitions as evolutionary reconfiguration processes: A multilevel perspective and a case-study. *Research Policy* 31(8/9): 1257-1274.
- Geels FW. (2005). System Innovation and the Transition to Sustainability. Theory, Evidence and Policy. Cheltenham: Edward Elgar.
- Geels, F.W., and J. Schot (2005). Taxonomy of sociotechnical transition pathways, unpublished paper.
- Gerlagh, R., and M.W. Hofkes (2002). Escaping Lock-in: the scope for a transition towards sustainable growth? Nota di Lavoro, 12.2002. Fondazione Eni Enrico Mattei, Milano, Italy.
- Gollier, C., B. Jullien and N. Treich (2000). Scientific progress and irreversibility: an economic interpretation of the precautionary principle. *Journal of Public Economics* 75: 229–253.
- Gowdy J. (1994) Coevolutionary Economics: The Economy, Society and the Environment. Dordrecht: Kluwer Academic Publishers.
- Gros, D. and A. Steinherr (2004) *Economic Transition in Central and Eastern Europe. Planting the Seeds*, Cambridge, Cambridge University Press.
- Greif, A. (1997). Cliometrics after 40 years. *American Economic Review* 87(2) Papers and Proceedings: 400-403.

- Harvey, M., and S. Metcalfe (2005). The Ordering of Change: Polanyi, Schumpeter and the Nature of the Market Mechanism, CRIC Discussion paper 70, March 2005.
- Hayami, Y. (1998). Development Economics. From Poverty to the Wealth of Nations, Oxford University Press, Oxford.
- Hayek, F.A. (1978). Competition as a Discovery Procedure, in *New Studies in Philosophy, Politics, Economics and the History of Ideas*, London, Routledge, 179-190.
- Henrich, J. (2004). Cultural group selection, coevolutionary processes and large-scale cooperation. *Journal of Economic Behavior and Organization* 53: 3-35.
- Hodgson, G.M. (1988). Economics and Institutions. A Manifesto for a Modern Institutional Economics. Cambridge, Oxford, UK, Polity Press, Blackwell Publisher.
- Hodgson, G.M., M. Itoh and N. Yokokawa (eds.) (2001). Capitalism in Evolution: Global Contentions East and West. Edward Elgar, Cheltenham.
- Holling, C.S. (1973). Resilience and stability of ecological systems. *Annual Review of Ecological Systems* 4: 1-24.
- Holling, C.S. (1986). The resilience of terrestrial ecosystems: Local surprise and global change.In: W. C. Clark and R. E. Munn (ed.). Sustainable Development of the Biosphere. Cambridge University Press, Cambridge, UK.
- Hollingsworth, J.R., and R. Boyer (1997). Contemporary Capitalism. The Embeddedness of Institutions. Cambridge University Press, Cambridge.
- Hubacek, K., and J.C.J.M. van den Bergh (2005). Changing concepts of 'land' in economic theory: from single to multi-disciplinary approaches. *Ecological Economics* 56(1): 5-27.
- Jacobsson, S., and A. Bergek, A. (2003). Transforming the energy system: The evolution of technological systems in renewable energy technology. In: K. Jacob, M. Binder and A. Wieczorek (eds.) (2004). *Governance for Industrial Transformation*. Proceedings of the 2003 Berlin Conference on the Human Dimensions of Global Environmental Change, Environmental Policy Research Centre, Berlin. pp. 208-236.
- Jaffe, A.B., M. Trajtenberg, and R. Henderson (1993). Geographic localization of knowledge spillovers as evidenced by patent citations. *Quarterly Journal of Economics* 108 (3): 577-598.
- Kahneman, D., and A. Tversky (1979). Prospect theory: an analysis of decision under risk. *Econometrica* 47: 263-291.
- Kemp, R., and L. Soete (1992) The greening of technological progress: An evolutionary perspective. *Futures* 24(5): 437-457.
- Kemp, R. (1994). Technology and the Transition to Environmental Sustainability. The Problem of Technological Regime Shifts. *Futures* 26(10): 1023-46.
- Kornai, J. (2005). The great transformation of central eastern Europe: Successes and disappointment, Presidential Address, delivered at the 14th World Congress of the International Economic Association in Marrakech, Morocco on August 29, 2005.
- Krugman, P. (1991b). Increasing returns and economic geography. *Journal of Political Economy* 99: 483-499.
- Kurgman, P. (1991) History versus expectations. Quarterly Journal of Economics, 106: 651-667.
- Kuznets, S. (1955). Economic growth and income inequality. *American Economic Review* 45(1): 1-28.
- Landes, D.S. (1994). What room for accident in history? Explaining big changes by small events, *Economic History Review* 19(4): 637-656.
- Layard, R. (2005). Happiness: Lessons from A New Science. Penguin, London.
- Levin et al. (17 authors) (1998). Resilience in natural and socioeconomic systems. Environment and Development Economics 3(2): 222-235.
- Malecki, E. (1997). Technology and Economic Development: The Dynamics of Local, Regional and National Competitiveness. 2<sup>nd</sup> edition. Addison Wesley Longman, Boston.

- Mäler, K.-G., A. Xepapadeas, and A.J. de Zeeuw (2003). The economics of shallow lakes. *Environmental and Resource Economics* 26(4): 603-624.
- Malerba, F, R. Nelson, L. Orsenigo and S. Winter, (2001) History-friendly models: An overview of the case of the computer industry. *Journal of Artificial Societies and Social Simulation*. 4(3): .
- Martin, S. (2001). Advanced Industrial Economics. Blackwell Publishing, Oxford.
- Molina, F., and L. Orsenigo (1997). Technological regimes and sectoral patterns of innovative activities. *Industrial and Corporate Change* 6(1): 83-117.
- Marx, K.Maynard Smith, J., and E. Szathmáry (1995). *The Major Transitions in Evolution*. Oxford University Press, Oxford.
- Mokyr, J. (1990). *The Lever of Riches: Technological Creativity and Economic Progress*. Oxford University Press, New York.
- Molina, A. (1999). Understanding the role of the technical in the buildup of sociotechnical constituencies. *Technovation* 19: 1-29.
- Nelson, R.R. (1990). Capitalism as an engine of progress. Research Policy 19(3): 193-214.
- Nelson, R.R. (1994). The Co-evolution of technology, industrial structure, and supporting institutions. *Industrial and Corporate Change* 3:47-64.
- Nelson R.R. and Winter S.G. (1982). An Evolutionary Theory of Economic Change. Bellknap Press, Cambridge, Mass.
- Norgaard, R.B. (1984). Coevolutionary development potential. Land Economics 60: 160-173.
- Norgaard, R.B. (1994). Development Betrayed: The End of Progress and a Coevolutionary Revisioning of the Future. Routledge, London and New York.
- North, D.C. (1981). Structure and Change in Economic History. W.W. Norton, New York.
- North D.C. (1991). Institutions, Institutional Change and Economic Performance. Cambridge University Press, Cambridge.
- North, D.C. (1997). Cliometrics 40 years later. American Economic Review 87(2): 412-414.
- Norton, B., R. Costanza and R.C. Bishop, 1998. The evolution of preferences. Why 'sovereign' preferences may not lead to sustainable policies and what to do about it. *Ecological Economics* 24: 193-211.
- Ostrom, E. (1990). Governing the Commons: The Evolution of Institutions for Collective Action New York. Cambridge University Press, Cambridge.
- Penn, D.J. (2003). The evolutionary roots of our environmental problems: towards a Darwinian ecology. *The Quarterly Review of Biology* 78(3): 275-301.
- Phillips, P.C.B. and D. Sul (2005). Economic transition and growth. Cowles Foundation Discussion Paper No. 1514, Yale University, New Haven.
- Pimm, S.L. (1984). The complexity and stability of ecosystems. Nature 307: 321-326.
- Pindyck, R.S. (2000). Irreversibilities and the timing of environmental policy. *Resource and Energy Economics* 22: 233-259.
- Potts, J. (2000). *The New Evolutionary Microeconomics: Complexity, Competence, and Adaptive Behavior*. Edward Elgar, Cheltenham.
- Raskin, Paul, Tariq Banuri, Gilberto Gallopin, Pablo Gutman, Al Hammond, Robert Kates, and Rob Swart (2002). *Great Transition. The Promise and Lure of the Times Ahead*, Tellus Institute, Boston.
- RMNO (2003). Omgaan met kennis in transities. Een verkenning van het ongekende. Raad voor Milieu- en Natuuronderzoek, Den Haag.
- Rosenkopf, L., and M.L. Tushman (1998). The coevolution of community networks and technology: Lessons from the flight simulation industry. *Industrial and Corporate Change*, 7: 311-346.
- Rosser, J.B. (1999). On the complexities of complex economic dynamics. *Journal of Economic Perspectives*, 13(4): 169-192.

- Rosser J.B., and M.V. Rosser (2004). Complex dynamics of macroeconomic complex and its aftermath in transition economies. *Eastern Ecomnomic Journal* 30(2): 207-221.
- Rostow, W. (1960). The Stages of Growth: A Non-communist Manifesto. Cambridge University Press, Cambridge.
- Rotmans, J. (2003). Transitiemanagement: Sleutel naar een Duurzame Samenleving. van Gorcum Uitgeverij, Assen.
- Schumpeter, J.A. (1934). The Theory of Economic Development. Harvard University Press, Cambridge, Mass.
- Schumpeter, J.A. (1939). Business Cycles: A Theoretical, Historical and Statistical Analysis of the Capitalist Process. 2 vols. McGraw-Hill, New York.
- Schumpeter, J.A. (1942). *Capitalism, Socialism and Democracy*. Harper and Brothers Publishers, New York.
- Siebert, H. (1985). Spatial aspects of environmental economics. In: A.V. Kneese and J.L. Sweeney (eds.). *Handbook of Natural Resource and Energy Economics*, Vol. 1. North-Holland, Amsterdam.
- Smith, A., A. Stirling and F. Berkhout (2005). The governance of sustainable socio-technical transitions. *Research Policy* 34(10): 1491-1510.
- Solow, R.M. (1956). A contribution to the theory of economic growth. *Quarterly Journal of Economics* 70: 65-94.
- Teece, D.J. (1986). Profiting from technological innovation Implications for integration, collaboration, licensing and public policy. *Research Policy* 15(6): 285-305.
- Tiebout, C. (1956). A pure theory of local expenditures. *Journal of Political Economy* 64: 416-424.
- Todaro, M.P., and S.C. Smith (2003). *Economic Development*. 8<sup>th</sup> edition. Pearson and sley, Boston.
- Tushman, M.L., and L. Rosenkopf (1992). Organizational determinants of technological advance: towards a sociology of technological evolution. *Research in Organizational Behaviour* 14: 311-347.
- Unruh, G. (2000). Understanding carbon lock-in. Energy Policy, 28: 817-830.
- Unruh, G. (2002). Escaping carbon lock-in. Energy Policy, 30: 317-325
- van den Heuvel, S.T.A., and J.C.J.M. van den Bergh (2005). Balancing diversity and efficiency: multilevel evolutionary dynamics in the solar photovoltaic industry. Working paper, FEWEB, Free University, Amsterdam.
- van Praag, B., and A. Ferrer-i-Carbonell (2004). *Happiness Quantified: A Satisfaction Calculus Approach*. Oxford University Press, Oxford.
- Verspagen, B., (1991). A new empirical approach to catching up or falling behind. *Structural Change and Economic Dynamics* 2(2): 359-380.
- Wilkinson, R. (1973). *Poverty and Progress: An Ecological Model of Economic Development.*, Methuen and Co., London.
- Wilson, D.S. (2002). Darwin's Cathedral: Evolution, Religion and the Nature of Society. University of Chicago Press, Chicago.
- Winder, N., B.S. McIntosh and P. Jeffrey (2005). The origin, diagnostic attributes and practical application of co-evolutionary theory. *Ecological Economics* 54(4): 347-361.
- Windrum, P., and C. Birchenhall (2005). Structural Change in the presence of network externalities: a co-evolutionary model of technological successions. *Journal of evolutionary economics* 15: 123-148.

# The UNU-MERIT WORKING Paper Series

- # 2006-001 A Knowledge Economy Paradigm and its Consequences by Luc Soete.
- # 2006-002 Public-Private Sector Partnerships in an Agricultural System of Innovation: Concepts and Challenges by Andy Hall.
- # 2006-003 Capacity Development for Agricultural Biotechnology in Developing Countries: Concepts, Contexts, Case Studies and Operational Challenges of a Systems Perspective by Andy Hall and Jeroen Dijkman.
- # 2006-004 Technological Capabilities with Different Degree of Coherence: A Comparative Study of Domestic-Oriented vs. Export-Driven Bulgarian Software Companies by Rossitza Rousseva.
- # 2006-005 Small Islands, New Technologies and Globalization: A Case of ICT adoption by SMEs in Mauritius by Kaushalesh Lal and Aveeraj Sharma Peedoly.
- # 2006-006 Beyond Unobserved Heterogeneity in Computer Wage Premiums; and Data on Computer use in Germany, 1997-2001. Double paper by Joan Muysken, Sybrand Schim van der Loeff and Valeria Cheshko.
- # 2006-007 Learning in Local Systems and Global Links: The Otigba Computer Hardware Cluster in Nigeria by Banji Oyelaran-Oyeyinka
- # 2006-008 Breaking the Fence: Patent Rights and Biomedical Innovation in "Technology Followers" by Padmashree Gehl Sampath
- # 2006-009 Taxation and Technolgoy Adoption: A Hotelling Approach by Ben Kriechel and Thomas Ziesemer
- # 2006-010 Foreign Direct Investment and Technology Spillovers: Evidence from the Indian Manufacturing Sector by Subash Sasidharan
- # 2006-011 Persistence of Innovation in Dutch Manufacturing: Is it Spurious? by Wladimir Raymond, Pierre Mohnen, Franz Palm and Sybrand Schim van der Loeff
- # 2006-012 Random walks and cointegration relationships in international parity conditions between Germany and USA of the post Bretton-Woods period by Franco Bevilacqua
- # 2006-013 On the Persistence of Inequality in the Distribution of Personal Abilities and Income by Adriaan van Zon and Hannah Kiiver
- # 2006-014 Foreign Direct Investment, Firm-Level Capabilities and Human Capital Development: Evidence from Kenyan Manufacturing Industry by Geoffrey Gachino
- # 2006-015 The Determinants of Pharmaceutical R&D Expenditures: Evidence from Japan by Jörg C. Mahlich and Thomas Roediger-Schluga
- # 2006-016 Random walks and cointegration relationships in international parity conditions between Germany and USA for the Bretton-Woods period by Franco Bevilacqua
- # 2006-017 Concepts and guidelines for diagnostic assessments of agricultural innovation capacity by Andy Hall, Lynn Mytelka and Banji Oyeyinka
- # 2006-018 Buying and Selling Research and Development Services, 1997 to 2002 by Julio M. Rosa, Antoine Rose and Pierre Mohnen

- # 2006-019 INDIA's product patent protection regime: Less or more of "pills for the poor"? by Padmashree Gehl Sampath
- # 2006-020 Worker Remittances and Growth: The Physical and Human Capital Channels by Thomas Ziesemer
- # 2006-021 Creating the Capacity to Benefit from Technological Change in Developing Countries by Watu Wamae
- # 2006-022 A Technology Acquisition Model: the role of learning and innovation by Watu Wamae
- # 2006-023 Are North-South Technological Spillovers Substantial? A dynamic panel data model estimation by Watu Wamae
- # 2006-024 How "black" is the black sheep compared to all others? Turkey and the EU by İ. Semih Akçomak and Saeed Parto
- # 2006-025 Diffusion paths for Micro Cogeneration Using Hydrogen in the Netherlands by Mattijs Taanman, Arend de Groot, René Kemp, & Bart Verspagen.
- # 2006-026 Rough Road to Market: Institutional Barriers to Innovations in Africa by Banji Oyelaran-Oyeyinka and Padmashree Gehl Sampath
- # 2006-027 Innovativity: a comparison across seven European countries by Pierre Mohnen, Jacques Mairesse & Marcel Dagenais
- # 2006-028 *Empirical studies of innovation in the knowledge driven economy* by Bronwyn H. Hall and Jacques Mairesse
- # 2006-029 Information and Intellectual Property: The Global Challenges by Rishab Ghosh and Luc Soete
- # 2006-030 Absorptive Capacity and Export Diversification in Sub-Saharan African Countries by Alexis Habiyaremye & Thomas Ziesemer
- # 2006-031 Indian Pharma Within Global Reach? By Padmashree Gehl Sampath
- # 2006-032 Foreign Ownership, Technological Capabilities and Exports: Evidence from 205 Clothing Firms in Sri Lanka by Ganeshan Wignaraja
- # 2006-033 (S,s) Pricing: Does the Heterogeneity Wipe Out the Asymmetry on Micro Level? by Zakaria Babutsidze
- # 2006-034 Brief History of Mobile Telecommunication in Europe by Theo Dunnewijk & Staffan Hultén
- # 2006-035 Irreversible Investment under Uncertainty in Electricity Generation: A Clay-Clay-Vintage Portfolio Approach with an Application to Climate Change Policy in the UK by Adriaan van Zon & Sabine Fuss

- # 2006-036 The structure of R&D collaboration networks in the European Framework Programmes by Thomas Roediger-Schluga and Michael J. Barber
- # 2006-037 What drives productivity in Tanzanian manufacturing firms: technology or institutions? By Micheline Goedhuys, Norbert Janz & Pierre Mohnen
- # 2006-038 Economics and Transitions: Lessons from Economic Sub-disciplines by Jeroen van den Bergh & René Kemp